\documentclass[draft]{adm-paper}
\usepackage{amsfonts,amsmath,amsthm,amscd,amssymb,latexsym}

\usepackage[a4paper]{geometry}

\usepackage{a4wide}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{graphicx}
\usepackage{tikz}
\usepackage{float}
\input xy
\xyoption{arrow}
\xyoption{matrix}

\def\WD{{\rm wd }\,}
\def\sdim{ {\rm sim.dim}\,}
\def\fdim{ {\rm fil.dim}\,}
\def\lfdim{ {\rm l.fil.dim}\,}
\def\gdim{{\rm gl.dim}\,}
\def\lgd{{\rm lgd}\,}
\def\gld{ {\rm gld}\,}
\def\lgld{ {\rm lgd}\,}
\def\GK{{\rm  GK}\,}
\def\Kdim{{\rm K.dim }\,}
\def\height{{\rm height }}
\def\Hom{{\rm Hom}}
\def\der{\partial }
\def\deri{{\partial }_i}
\def\derj{{\partial }_j}
\def\nFu{{\nu }_{F,u}}
\def\nFm{{\nu }_{F,m}}
\def\nGn{{\nu }_{G,n}}
\def\nFM0{{\nu }_{F,M_0}}
\def\nFN0{{\nu }_{F,N_0}}
\def\nGN0{{\nu }_{G,N_0}}
\def\nFv{{\nu }_{F,v}}
\def\nF{{\nu }_{F}}
\def\nGv{ {\nu }_{G,v} }
\def\nG{ {\nu }_{G} }
\def\Nat{{\bf N}}
\def\N0{ {\bf N}_0 }
%\def\R{{\bf R }}
\def\R{\mathbb{R}}
\def\Rp{{\bf R }'}
\def\t{\otimes}
\def\g{\gamma}
\def\v{\varphi}
\def\ra{\rightarrow}
\def\la{\leftarrow}
\def\lra{\leftrightarrow}
\def\Xpm{X^{\pm }}
\def\Vpm{V_{\pm }}
\def\Cpmo{C^{\pm 1 }}
\def\Zpmo{Z^{\pm 1 }}
\def\Hpmo{H^{\pm 1}}
\def\HQ{{\cal H}_q}
\def\chij{ {\chi }_j}
\def\chik{ {\chi }_k}
\def\s{\sigma}
\def\Z{\mathbb{Z}}
\def\lk{{\lambda}_k}
\def\lj{{\lambda}_j}
\def\l1{{\lambda}_1}
\def\ls{{\lambda}_s}

\def\li{{\lambda}_i}
\def\gtk{ g_{t(k)}}
\def\p{{\bf p}}
\def\m{{\bf m}}
\def\a{\alpha}
\def\a0{ {\alpha }_0}
\def\a1{ {\alpha }_1}
\def\ai{{\alpha}_{i}}
\def\ak{{\alpha}_{k}}
\def\al{a_{\lambda }}
\def\as{ {\alpha }_s}
\def\am{{\alpha }_m}
\def\at{{\alpha }_t}

\def\l{\lambda}
\def\o{\omega}
%%%%   2   %%%%%%

\def\rfd{ {\rm rfd}\,}
\def\rfdim{ {\rm rel.f.dim}\,}
\def\nFGM0{{\nu }_{F,G,M_0}}
\def\nFG{{\nu }_{F,G}}
%%%%%%%%%%%%%%%%%%%%%%    section    3   %%%%%%%%%%%%%%%%%%%%%%%%%%%%
\def\lc{{\rm lc}}
\def\nFN0{{\nu}_{F,N_0}}
\def\tnu{\tilde{\nu}}
\def\lab{{\lambda }_{\alpha \beta }}
\def\Xa{X^{\alpha }}
\def\db{{\partial}^{\beta}}
\def\tnu{\tilde{\nu }}

%%%%%%%%%%%%%%%%%%%%%      section    4        %%%%%%%%%%%%%%
\def\GWA{D(\sigma ,a)}
\def\sn{{\sigma}^n}
\def\sm{{\sigma}^m}
\def\si{{\sigma}^i}
\def\sj{{\sigma}^j}
\def\sk{{\sigma}^k}
\def\sni{{\sigma}^{n-i}}
\def\st{{\sigma}^t}

\def\snm{{\sigma}^{n-m}}
\def\smm{{\sigma}^{-m}}
\def\smi{{\sigma}^{-i}}
\def\smj{{\sigma}^{-j}}
\def\smt{{\sigma}^{-t}}
\def\smk{{\sigma}^{-k}}
\def\sms{{\sigma}^{-s}}
\def\sm1{{\sigma}^{-1}}
\def\smti{{\sigma}^{-t_i}}
\def\smts{{\sigma}^{-t_s}}
\def\smr{{\sigma}^{-r}}
\def\smtp1{{\sigma}^{-t+1}}
\def\smtk{{\sigma}^{-t_k}}


\def\siZ{{\sum}_{i\in \mathbb{Z}}\,}
\def\o{\omega }
\def\S1{S^{-1}}
\def\SP{S^+}
\def\SPM{S^{\pm }}
\def\SM{S^-}
\def\ScD{S_{\cal D}}
\def\Xpm1{X^{\pm 1}_1}
\def\XM{X^-}
\def\XP{X^+}
\def\XPM{X^{\pm }}
\def\XMP{X^{\mp }}
\def\Xpmn{X^{\pm 1}_n}
\def\dPM{{\partial }^{\pm }}
\def\dP{{\partial }^+}
\def\sPM1{{\sigma }^{\pm 1}}
\def\sMP1{{\sigma }^{\mp 1 }}
\def\RPM{R^{\pm }}

\def\skI{{\sigma}^k_i}
\def\skmI{{\sigma}^{k-m}_i}
\def\smmI{{\sigma}^{-m}_i}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\def\b{\beta}
\def\d{\delta}
\def\dk{{\delta}_k}
\def\q{{\bf q}}
\def\aa{{\bf a}}
%\def\pd{{\rm pd}_A}
\def\pdab{{\rm pd}_{A\otimes B}}
\def\pdb{{\rm pd}_B}
\def\pdd{{\rm pd}_D}
\def\fd{{\rm fd}_A}
\def\pdb{{\rm pd}_B}
\def\fdb{{\rm fd}_B}
\def\cc{{\bf c}}
\def\bb{{\bf b}}
\def\AAA{{\bf A}}
\def\BB{{\bf B}}
\def\di{{\rm d.ind}}
\def\PI{{\prod}_{\alpha \in I}}
\def\MA{M_{\alpha}}
\def\LA{L_{\alpha}}
\def\NA{N_{\alpha}}
\def\AA{A_{\alpha}}
\def\SA{S_{\alpha}^{-1}}
\def\RM{R_{{\bf m}}}
\def\M{{\cal M}}
\def\L{\Lambda}
\def\O{\Omega}
\def\LI{\Lambda {}_i}
\def\AH{A=K[H](\sigma ,a)}
\def\G{\Gamma}
\def\ta{\otimes {}_A}
\def\td{\otimes {}_D}
\def\TL{{\otimes}^n_{i=1}\Lambda {}_i}
%\def\SL{\sum {}^n_{i=1}{\rm d}\Lambda {}_i}
\def\SLL{\sum {}^n_{i=1}{\rm lgd}\Lambda {}_i}
\def\SLW{\sum {}^n_{i=1}{\rm wd}\Lambda {}_i}
\def\A{A=D(\sigma,a)}
\def\UM{\cup {}_{i\in I}M_i}
\def\UN{\cup {}_{i\in I}N_i}
\def\LL{L_i/L_{<i}}
\def\NN{N_i/N_{<i}}
\def\OO{{\cal O}}
\def\CA{{\cal A}}
\def\CAP{{\cal A}^+}
\def\CAM{{\cal A}^-}
\def\CAPM{{\cal A}^{\pm }}
\def\cFN{{\cal FN}}
\def\cCN{{\cal CN}}
\def\cLN{{\cal LN}}
\def\CD{{\cal D}}
\def\CDPM{{\cal D}^{\pm }}
\def\OP{{\cal O}({\bf p})}
\def\LLV{\tilde{L_i}/\tilde{L_{<i}}}
%%%%%%%%%%%%%%%%%%%%% glgwa %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\def\TPM{T^{\pm }}
\def\ATp{A_{T^+}}
\def\ATm{A_{T^-}}
\def\ATPM{A_{T^{\pm }}}
\def\ASP{A_{S^+}}
\def\ASM{A_{S^-}}

\def\ASPM{A_{S^{\pm }}}
\def\LPol{D[X,X^{-1};\sigma ]}
\def\chainM{M=M_n\supseteq M_{n-1}\supseteq \ldots \supseteq M_0=0}
\def\MiM{M_i/M_{i-1}}
\def\ppi{{\bf p}_i}
\def\OOle{{\cal O}_{\le }}
\def\annD{{\rm ann}_D\,}
\def\AM{ {}_AM}
\def\Yti{Y^{t_i}}
\def\Ytk{Y^{t_k}}
\def\Ytr{Y^{t_r}}
\def\Ytm1{Y^{t-1}}
\def\Yim1{Y^{i-1}}
\def\Dp{D_{\bf p}}
\def\pmu{{\mu }\prime}
\def\vxi{\vec {\xi }}
\def\r{{\bf r}}
\def\txi{\tilde {\xi }}

%%%%%%%%%%%%%%%%%%%%%  kgwa   %%%%%%%%%%%%%%%%%%%%%%%%%%%
\def\CK{{\cal K}}
\def\CL{{\cal L}}
\def\CM{{\cal M}}
\def\CN{{\cal N}}
\def\CS{{\cal S}}
\def\CF{{\cal F}}
\def\CG{{\cal G}}
\def\CH{{\cal H}}
\def\ass{{\rm ass}}
\def\bT{\overline{T}}

\def\ZN{\mathbb{Z}^n}
\def\Zn{\mathbb{Z}_n}
\def\k{{\bf k}}
\def\e{{\bf e}}
\def\i{{\bf i}}


%%%%%%%%%%%%%%  dix  %%%%%%%%%%%%
\def\CZ{{\cal Z}}
\def\supp{{\rm supp }}
\def\CFN{{\cal F}_{\bf N}}
\def\Zn{\mathbb{Z}_n}
\def\HN{H_{\bf N}}
\def\EN{E_{\bf N}}
\def\GN{G_{\bf N}}
\def\Aut{{\rm Aut}}
\def\bK{\overline{K}}
\def\bQ{\overline{Q}}
\def\bA{\overline{A}}
\def\Der{{\rm Der }}
\def\ad{{\rm ad }}
\def\dim{{\rm dim }}
\def\char{{\rm char }}
\def\pad{{\rm pad }}
\def\LD{{\rm LD }}
\def\Iso{{\rm Iso }}
\def\Isomax{{\rm Iso.max }}
\def\bi{\bar{i}}
\def\bj{\bar{j}}


%%%%%%%%%%%%%%%  dixpr5, jormod %%%%%%%%%%%%%%%%%%%%%%%%%
\def\ker{ {\rm ker } }
\def\ndeg{ {\rm ndeg} }
\def\CJ{ {\cal J}}
\def\gr{ {\rm gr} }
\def\gcd{ {\rm gcd } }
\def\D{ \Delta }
%%%%%%%%%%%%%%%%%%%%  dixpr3   %%%%%%%%%%%
\def\Ev{ {\rm Ev} }

%%%%%%%%%%%%%%%%%%%%  dixpr2   %%%%%%%%%%%
\def\Ali{A_{\leq i}}
\def\Agi{A_{\geq i}}
\def\Conv{{\rm Conv }}
\def\mm{ {\bf m} }
\def\xx{ {\bf x} }
\def\yy{ {\bf y} }
\def\zz{ {\bf z} }
\def\SLnZ{ {\rm SL}_n(\mathbb{Z}) }
\def\SL2Z{ {\rm SL}_2({\bf Z}) }
\def\ll{ \overrightarrow{\lambda } }
\def\nn{ {\bf n} }
\def\CZ{ {\cal Z}}
\def\CR{ {\cal R}}
%\def\CZQ{ {\cal Z}_\mathbb{Q}}
%\def\CFQ{ {\cal F}_\mathbb{Q}}
%\def\Zs{ {\mathbb{Z}^s }
%\def\Qs{ {\mathbb{Q}^s }
\def\LD{ {\rm LD} }
\def\lcmden{ {\rm lcm.den } }
\def\lcmnom{ {\rm lcm.nom } }
\def\th{ \theta }
\def\Esr{E_{\sigma , \rho }}
\def\CLsr{ {\cal L}^{\sigma , \rho } }
\def\CLuu{ {\cal L}^{-1 , 1 } }
\def\CLou{ {\cal L}^{0 , 1 } }
\def\hCLou{ \widehat{\cal L}^{0 , 1 } }
\def\CL{{\cal L}}

\def\Lsr{L^{\sigma , \rho } }
\def\Psr{P^{\sigma , \rho } }
\def\Gsr{G^{\sigma , \rho } }
\def\Gsrpp{G^{\sigma' , \rho' } }
\def\Gp1{ G^{1 , 1 } }
\def\P11{ P^{-1 , 1 } }
\def\Pp1{ P^{1 , 1 } }
\def\Pou{ P^{0 , 1 } }
\def\vsr{v_{\sigma , \rho }}
\def\lsr{l_{\sigma , \rho }}
\def\rsr{r_{\sigma , \rho }}
\def\wsr{w_{\sigma , \rho }}
\def\bP{ \bar{P}}
\def\Paut{{\rm Paut }}
\def\Pautgr{{\rm Paut }_{gr}}
\def\Pautgr{{\rm Paut }_{gr}}
\def\Pend{{\rm Pend}}
\def\Autgr{{\rm Aut }_{gr}}
\def\hst{-\frac{\xi }{(\s +\rho )\omega L(\alpha )}\, yx}
\def\Exv{{\rm Exv}}
\def\Exs{{\rm Exs}}
\def\BV{{\rm BV }}
\def\PE{{\rm PE }}
\def\TE{{\rm TE }}
\def\TV{{\rm TV }}
\def\RE{{\rm RE }}
\def\ux{\underline{x}}
\def\NP{{\rm NP}}
\def\Supp{{\rm Supp}}
\def\RT{{\rm RT}}
\def\lte{{\rm lte}}
\def\vol{{\rm vol}}
\def\cont{{\rm cont}}
\def\rdeg{{\rm r.deg}}
\def\vt{\vartheta}
\def\lcm{{\rm lcm }}
\def\th{\theta}
\def\CE{{\cal E}}
\def\CV{{\cal V}}
\def\CVhom{{\cal V}^{hom}}
\def\CEhom{{\cal E}^{hom}}
\def\Cspr{C_{\sigma +\rho}}

%%%%%%%%%%%%%   jc2   %%%%%%%%%%%%
\def\nCLsr{{}^\nu\kern-2pt {\cal L}^{\sigma , \rho  }}
\def\nP{{}^\nu \kern-2pt P}
\def\nL{{}^\nu\kern-2pt L}
\def\nLL{{}^\nu\kern-2pt \Lambda}
\def\nPsr{{}^\nu\kern-2pt P^{\sigma , \rho  }}
\def\nLsr{{}^\nu\kern-2pt L^{\sigma , \rho  }}
\def\nuCL{{}^\nu\kern-2pt  {\cal L}}
\def\nCLsr{{}^\nu\kern-2pt {\cal L}^{\sigma , \rho  }}
\def\nCL1m{{}^\nu\kern-2pt {\cal L}^{-1 , 1  }}
\def\x1nu{x^\frac{1}{\nu}}
\def\xm1nu{x^{-\frac{1}{\nu}}}
\def\xinu{x^\frac{i}{\nu}}
%\def\CNNN#1#2{\phantom{|}_{{#1}} {\Cal N}_{#2}}
%\def\N2#1#2{\phantom{|}_{{#1}} {N}_{#2}}


%%%%%%%%%%%%%%%  pa4     %%%%%%%%%%%
\def\rdeg{{\rm rdeg}}
\def\rad{{\rm rad}}
\def\EP{{\cal EP}}
%%%%%%%%%%%   luroth  %%%%%%%%%%
\def\trdeg{{\rm tr.deg}}
\def\bF{\overline{F}}
\def\tK{\widetilde{K}}
\def\ob{\overline{b}}
\def\bX{{\bf X}}

%%%%%%%%%%%%%%%    dmodbook  %%%%%%%%%%%%%%%
\def\CR{ {\cal R}}
\def\hAn{ \hat{A_n}}
\def\DF{{\cal Definition}.}
\def\EX{{\cal Example}.}
\def\bL{\bar{L}}
\def\CN{{\cal N}}
\def\ra{\rightarrow }
\def\SPR{R[t; \sigma, \delta ]}
\def\CB{{\cal B}}
\def\lcm{{\rm lcm}}
\def\CI{{\cal I}}
\def\CCd{{\cal C}_.}
\def\coker{{\rm coker}}
\def\CT{{\cal T}}

%%%%%%%%%%%%   fdim   %%%%%%%%%%%%%%%%%

\def\CC{ {\cal C}}
\def\CE{ {\cal E} }
\def\CH{ {\cal H}}
\def\CP{ {\cal P}}
\def\Ao{ A^{\rm o} }
\def\Bo{ B^{\rm o} }
\def\Ae{ A^{\rm e} }
\def\CAe{ {\cal A}^{\rm e} }
\def\CAo{ {\cal A}^{\rm o} }
\def\nAM0{{\nu }_{{\cal A},M_0}}
\def\nAN0{{\nu }_{{\cal A},N_0}}
\def\hol{ {\rm hol } }
\def\Vect{ {\rm Vect } }
\def\Kdim{ {\rm Kdim } }
\def\lfdim{ {\rm lfdim } }
%%%%%%%%%%%%%%%%%%%%%  jcdp    %%%%%%%%%%%%%%%%%
\def\End{ {\rm End }}
\def\Der{ {\rm Der }}
\def\CJ{ {\cal J }}
\def\CR{ {\cal R }}
\def\CP{ {\cal P }}
\def\det{ {\rm det }}
\def\ad{ {\rm ad }}

%%%%%%%%%%%%%%%    waclnd    %%%%%%
\def\hCA{ \widehat{\CA} }

%%%%%%%%%%%%%%%%%  diff  %%%%%%
\def\EKP{ {\rm End}_K(P)}
\def\EKPI{ {\rm End}_K(P/I)}
\def\hAX{\widehat{A}_X}
\def\tAXI{ \widetilde{A}_{X, I}}
\def\AXI{A_X^I}
\def\AnI{A_n^I}
\def\AmJ{A_m^J}
\def\AnmF{A_{n+m}^F}
\def\AXIi{A_{X,i}^I}
\def\AnIi{A_{n,i}^I}
\def\tCD{\widetilde{\cal D}}
\def\NL{\mathbb{N}^{(\Lambda )}}
\def\bI{\overline{I}}
\def\bp{\overline{p}}
\def\bR{\overline{R}}
\def\spec{{\rm spec}}
\def\tr{{\rm tr}}
\def\rl{{\rm rl}}
\def\lrl{\underline{{\rm rl}}}
\def\ol{{\rm ol}}
\def\bQ{\overline{Q}}
\def\bQn{\overline{Q}_n}
\def\bx{\overline{x}}
\def\bz{\overline{z}}
\def\by{\overline{y}}
\def\bq{\overline{q}}
\def\bw{\overline{w}}
\def\of{\overline{f}}
\def\oX{\overline{X}}
\def\bMI{\overline{\mathbb{I}}}
\def\bxl{\overline{x}_l}
\def\PS{P^{(S)}}
\def\PT{P^{(T)}}

%%%%%%  fgdiva  %%%%%%%%%
\def\Gg{\mathfrak{g}}
\def\ga{\mathfrak{a}}
\def\gb{\mathfrak{b}}
\def\gc{\mathfrak{c}}
\def\gd{\mathfrak{d}}
\def\ge{\mathfrak{e}}
\def\gn{\mathfrak{n}}
\def\gm{\mathfrak{m}}
\def\gp{\mathfrak{p}}
\def\gq{\mathfrak{q}}
\def\gr{\mathfrak{r}}
\def\gA{\mathfrak{A}}
\def\gB{\mathfrak{B}}
\def\gC{\mathfrak{C}}
\def\gD{\mathfrak{D}}
\def\gE{\mathfrak{E}}
\def\gN{\mathfrak{N}}
\def\gM{\mathfrak{M}}
\def\gP{\mathfrak{P}}
\def\gQ{\mathfrak{Q}}
\def\gR{\mathfrak{R}}
\def\gder{\gd\ge\gr}
%%%%%%%%%% gendifreg  %%%%%%%%
\def\bJ{\overline{J}}
\def\tb{\widetilde{b}}
\def\tB{\widetilde{B}}
\def\hj{\widehat{j}}
\def\hi{\widehat{i}}
\def\dij{\delta_{{\bf i}, {\bf j}}}
\def\vij{v_{{\bf i}, {\bf j}}}
\def\tvij{\widetilde{v_{{\bf i}, {\bf j}}}}
\def\vijnu{v_{{\bf i}, j_\nu}}
\def\tvijnu{\widetilde{v}_{{\bf i}, j_\nu}}
\def\vijmu{v_{{\bf i}, j_\mu}}
\def\tvijmu{\widetilde{v}_{{\bf i}, j_\mu}}
\def\vijk{v_{{\bf i}, j_k}}
\def\tvijk{\widetilde{v}_{{\bf i}, j_k}}
\def\vijmus{v_{{\bf i}, j_\mu'}}
\def\tvijmus{\widetilde{v}_{{\bf i}, j_\mu'}}
\def\Dijnu{\Delta_{{\bf i}, j_\nu}}
\def\tDijnu{   \widetilde{\Delta}_{{\bf i}, j_\nu}  }
\def\Dijnus{\Delta_{{\bf i}, j_\nu'}}
\def\Dij{\Delta_{{\bf i}, {\bf j}}}
\def\Disj{\Delta_{{\bf i}', {\bf j}}}
\def\tDijnus{\widetilde{\Delta}_{{\bf i}, j_\nu'}}
\def\tDij{\widetilde{\Delta}_{{\bf i}, {\bf j}}}
\def\tDisj{\widetilde{\Delta}_{{\bf i}', {\bf j}}}
\def\Hij{H( {\bf i}, {\bf j})}
\def\Hjjs{H( {\bf j}, {\bf j}')}
\def\Hjsj{H( {\bf j}', {\bf j})}
\def\Hji{H( {\bf j}, {\bf i})}
\def\tHij{\widetilde{H}( {\bf i}, {\bf j})}
\def\tHjjs{\widetilde{H}( {\bf j}, {\bf j}')}
\def\tHjsj{\widetilde{H}( {\bf j}', {\bf j})}
\def\tHji{\widetilde{H}( {\bf j}, {\bf i})}
\def\derij{\partial_{{\bf i}, {\bf j}}}
\def\GL{{\rm GL}}
\def\SL{{\rm SL}}
\def\j{{\bf j}}
%\def\bJij{\bJ ({\bf i}, {\bf j})}
\def\bJisjs{\overline{J}({\bf i}', {\bf j}')}
\def\tJisjs{\widetilde{J}({\bf i}', {\bf j}')}
\def\tp{\widetilde{p}}
\def\Spec{{\rm Spec}}
\def\tdfixj{\widetilde{\frac{\der f_i}{\der x_j}}}
\def\taa{\widetilde{{\bf a}}}
\def\tga{\widetilde{\mathfrak{a}}}
\def\ddij{d_{{\bf i} , {\bf j}}}
\def\tD{\widetilde{\Delta}}
\def\tJ{\widetilde{J}}
\def\tpi{\widetilde{\pi}}
\def\tv{\widetilde{v}}
\def\tH{\widetilde{H}}
\def\tF{\widetilde{F}}
\def\tV{\widetilde{V}}
\def\tJ{\widetilde{J}}
\def\Ext{{\rm Ext}}
\def\Hom{{\rm Hom}}
\def\II{{\bf I}}
\def\JJ{{\bf J}}
\def\gder{\mathfrak{der}}

%%%%%%%%%% charpgendif   %%%%
\def\HS{{\rm HS}}
\def\di!{\frac{\der^i}{i!}}
\def\dik!{\frac{\der^k_i}{k!}}
%%%%% invendom   %%%%
\def\hA{\widehat{A}}
\def\hADi{\widehat{A}^{D_i}}
%%%%%%%%%%%%%%%%%%  genrelinv   %%%%
\def\Ns{\mathbb{N}^s}
\def\grA{{\rm gr}(A)}
\def\grphi{{\rm gr}(\phi)}
\def\grpsi{{\rm gr}(\psi)}
\def\zdai{z_{d , \alpha , i}}
\def\tdai{t_{d , \alpha , i}}
\def\ord{{\rm ord}}

%%%%%%%%%% invendom  %%%%%%%%%%
\def\bd{\overline{\delta}}
\def\Adhi{A^{\delta}_{\widehat{i}}}
\def\Adhj{A^{\delta}_{\widehat{j}}}
\def\tK{\widetilde{K}}
\def\Fp{\mathbb{F}_p}


%%%%%%%%%%%%% chpinv  %%%%%%%%%%%%%%
\def\xba{x^{[\alpha ]}}
\def\xbb{x^{[\beta ]}}
\def\xbi{x^{[i]}}
\def\xbj{x^{[j]}}
\def\ybi{y^{[i]}}
\def\ybj{y^{[j]}}
\def\xbij{x_i^{[j]}}
\def\xbik{x_i^{[k]}}
\def\xbji{x_j^{[i]}}
\def\derik{\der_i^{[k]}}
\def\derij{\der_i^{[j]}}
\def\derjk{\der_j^{[k]}}
\def\deril{\der_i^{[l]}}
\def\derjl{\der_j^{[l]}}
\def\derba{\der^{[\alpha ]}}
\def\derbb{\der^{[\beta ]}}
\def\dersba{{\der'}^{[\alpha ]}}
\def\dersbb{{\der'}^{[\beta ]}}
\def\Nn{\mathbb{N}^n}
\def\Nm{\mathbb{N}^m}
\def\Ns{\mathbb{N}^s}
\def\Pnhi{P_{n,\widehat{i}}}
\def\Ahr{A_{\widehat{r}}}
\def\hP{\widehat{P}}
\def\hgm{\widehat{\gm}}
\def\gl{\mathfrak{l}}
\def\NI{\mathbb{N}^{(I)}}
\def\id{{\rm id}}
%%%%%%%%%%%%%%%%%%%%%%  dsimnn.tex   %%%%%%%%%%%%%
\def\hT{\widehat{T}}
\def\hgm{\widehat{\gm}}

%%%%%%%%%%%%%%%%%%%%%%  simdernp.tex %%%%%%%%%%%%%%%%%%%%
\def\nsder{{\rm nsder}}
\def\sder{{\rm sder}}

%%%%%%%%%%%%%%%%%%%%%%  simderharp.tex %%%%%%%%%%%%%%%%%%%%
\def\ID{{\rm ID}}

%%%%%%%%%%%%%%%%%%%%   diflinep    %%%%%%%%%%%%%%
\def\Zp{\mathbb{Z}_p}
\def\Zpp{\mathbb{Z}_{(p)}}
\def\poi{{\rm poi}}
\def\Max{{\rm Max}}
\def\FpN{\mathbb{F}_p^\mathbb{N}}
\def\gldim{{\rm gldim}}
\def\SpN{S_p^\mathbb{N}}
\def\Fun{{\rm Fun}}

%%%%%%%%%%%%%%  autgras   %%%%%%%%%%%%%%%%%%
\def\N{\mathbb{N}}

\def\0{\overline{0}}
\def\1{\overline{1}}
\def\Lnz{\L_{n,\overline{0}}}
\def\Ln1{\L_{n,\overline{1}}}
\def\Lni{\L_{n,\overline{i}}}
\def\oa{\overline{a}}
\def\Lnod{\L_n^{od}}
\def\Lnev{\L_n^{ev}}
\def\Lnsod{\L_n'^{od}}
\def\Lnsev{\L_n'^{ev}}
\def\SDer{{\rm SDer}}
\def\SDerr{{\rm SDer}^r}
\def\Derod{\Der_K (\L_n)^{od}}
\def\Derev{\Der_K(\L_n)^{ev}}
\def\Derevs{\Der_K(\L_n)^{ev}_s}
\def\Derevs{\Der_K(\L_n)^{ev}_s}
\def\Derevn{\Der_K(\L_n)^{ev}_n}
\def\Derods{\Der_K(\L_n)^{od}_s}
\def\Derodn{\Der_K(\L_n)^{od}_n}
\def\SDerod{\SDer_K (\L_n)^{od}}
\def\SDerev{\SDer_K(\L_n)^{ev}}
\def\SDerrod{\SDer_K^r (\L_n)^{od}}
\def\SDerrev{\SDer_K^r(\L_n)^{ev}}
\def\IDer{{\rm IDer}}
\def\ISDer{{\rm ISDer}}
\def\uod{u^{od}}
\def\uev{u^{ev}}
\def\dod{\d^{od}}
\def\dev{\d^{ev}}
\def\az{a_{\overline{0}}}
\def\a1{a_{\overline{1}}}
\def\sad{{\rm sad}}
\def\bs{\overline{s}}
\def\St{{\rm St}}
\def\S{\Sigma}
\def\tS{\widetilde{\Sigma}}
\def\grad{{\rm grad}}
\def\bCJ{\overline{\CJ}}
\def\CU{{\cal U}}
\def\Stab{{\rm Stab}}
\def\hCJ{\overline{\CJ}}

%%%%%%%%%%%%%%%  monpa   %%%%%%%%%%%%%%%%%
\def\Mon{{\rm Mon}}
\def\Dec{{\rm Dec}}
\def\Sign{{\rm Sign}}
\def\sign{{\rm sign}}
\def\vn1{\overrightarrow{n-1}}
\def\CQ{{\cal Q}}
%%%%%%%%%%% autpol    %%%%%%%%%%%%%
\def\Sh{{\rm Sh}}
\def\Aff{{\rm Aff}}
\def\hx{\widehat{x}}
\def\hP{\widehat{P}}
\def\hPhi{\widehat{\Phi}}
\def\hS{\widehat{\Sigma}}
\def\hhS{\widehat{S}}
\def\hhG{\widehat{G}}
\def\hG{\widehat{\G}}
\def\hE{\widehat{E}}
\def\hD{\widehat{D}}
\def\hCT{\widehat{\CT}}
\def\hEs{\widehat{E}'}
\def\hPn{\widehat{P}_n}
\def\hEn{\widehat{E}_n}
\def\hEns{\widehat{E}_n'}
\def\hCP{\widehat{\CP}}
\def\hOO{\widehat{\OO}}
\def\Q{\mathbb{Q}}
\def\im{{\rm im}}
\def\Gr{{\rm gr}}
\def\gl{{\rm gl}}
\def\sl{{\rm sl}}
\def\Dim{{\rm Dim}}
\def\Af{{\rm Af}}
\def\Uf{{\rm Uf}}
\def\mA{\mathbb{A}}
\def\mD{\mathbb{D}}
\def\mL{\mathbb{L}}

%%%%%%%%%%%%%%%% jacalg   %%%%

\def\gf{\mathfrak{f}}
\def\soc{{\rm soc}}
\def\Sub{{\rm Sub}}
\def\SSub{{\rm SSub}}
\def\Inc{{\rm Inc}}
\def\Min{{\rm Min}}
\def\csupp{{\rm csupp}}


%%%%%%%%%%%%% Frobdifop  %%%%%%%%%%%%%

\def\DF{{\cal DF}}
\def\deripk{\der_i^{[p^k]}}
\def\derjpk{\der_j^{[p^k]}}
\def\deripl{\der_i^{[p^l]}}
\def\derjpl{\der_j^{[p^l]}}
\def\derijkpk{\der_i^{[j_kp^k]}}
\def\deripkjk{\der_i^{[p^k]j_k}}
\def\CDPn{\CD (P_n)}
\def\DFPn{\DF (P_n)}
\def\grCDPn{{\rm gr}_\CF (\CDPn )}
\def\grF{{\rm gr}(F)}
\def\bu{\overline{u}}
\def\PZ{{\rm PZ}}
\def\CPI{{\cal PI}}

%%%%%%%%%%%%%%%%  Frobext   %%%%%%%%%%%%%%%%%%
\def\FrobCDPn{{\rm Frob}_p(\CD (P_n))}
\def\FrobsCDPn{{\rm Frob}_{p,s}(\CD (P_n))}
\def\FrobsCDPnDn{{\rm Frob}_{p,s}(\CD (P_n), \D_n)}
\def\FrobCDPnDn{{\rm Frob}_p(\CD (P_n), \Delta_n))}
\def\FrobCDPnPn{{\rm Frob}_p(\CD (P_n), P_n))}
\def\FrobsCDPnPn{{\rm Frob}_{p,s}(\CD (P_n), P_n))}
\def\Frob{{\rm Frob}}
\def\F{\mathbb{F}}
\def\dps{\delta^{p^s}}
\def\ID{{\rm ID}}
\def\derispk{\der_i'^{[p^k]}}
\def\derips{\der_i^{[p^s]}}
\def\StDn{ {\rm St}(\D_n)}
\def\stDn{ {\rm st}(\D_n)}
\def\FPolCDPn{{\rm FPol}(\CDPn )}
\def\Inn{{\rm Inn}}

%%%%%%%%%%%%%%%%  Frobaut   %%%%%%%%%%%%%%%%%%
\def\CDQn{\CD (Q_n)}
\def\Qp{\mathbb{Q}_p}

%%%%%%%%%%%%%%%%  shrekalg   %%%%%%%%%%%%%%%%%%
\def\mS{\mathbb{S}}
\def\mJ{\mathbb{J}}
\def\mI{\mathbb{I}}
\def\mrD{\mathrm{D}}
\def\ann{{\rm ann}}
\def\lann{{\rm l.ann}}
\def\rann{{\rm r.ann}}
\def\Cen{{\rm Cen}}
\def\clKdim{{\rm cl.Kdim}}
\def\pd{{\rm pd}}
\def\hmS{\widehat{\mathbb{S}}}
\def\hht{{\rm ht}}
\def\cht{{\rm cht}}
\def\lgldim{{\rm l.gldim}}
\def\rgldim{{\rm r.gldim}}
\def\lwdim{{\rm l.wdim}}
\def\rwdim{{\rm r.wdim}}
\def\fd{{\rm fd}}
\def\bM{\overline{M}}


%%%%%%%%%%%%%%%%  shrekaut   %%%%%%%%%%%%%%
\def\heta{\widehat{\eta}}
\def\mF{\mathbb{F}}
\def\mM{\mathbb{M}}
\def\mT{\mathbb{T}}
\def\ind{{\rm ind}}
\def\ud{\underline{d}}
\def\stCH{{\rm st}_{G_n}(\CH_1)}


%%%%%%%%%%%%%%%%  shrekaut   %%%%%%%%%%%%%%
\def\mG{\mathbb{G}}
\def\mE{\mathbb{E}}
\def\mX{\mathbb{X}}
\def\mU{\mathbb{U}}
\def\Frac{{\rm Frac}}
\def\Out{{\rm Out}}

%%%%%%%%%%%%%%%  Snaut   %%%%%%%%%%%%%%
\def\bgp{\overline{\gp}}
\def\bTh{\overline{\Theta}}
%\def\K0{{\rm K}_0}
\def\K1{{\rm K}_1}
\def\bG{\overline{\G}}
\def\bpsi{\overline{\psi}}
\def\mK{\mathbb{K}}
\def\mP{\mathbb{P}}
\def\mY{\mathbb{Y}}
\def\mA{\mathbb{A}}
\def\mB{\mathbb{B}}
\def\SAff{{\rm SAff}}
\def\aff{{\rm aff}}

%%%%%%%%%%%  K1-Sn-group   %%%%%%%%%%

\def\tTh{\widetilde{\Theta}}
\def\tmE{\widetilde{\mathbb{E}}}
\def\bdet{\overline{\det}}


%%%%%%%%%%%  HilbSn   %%%%%%%%%%

\def\wdim{{\rm w.dim}}
\def\btaut{\overline{\tau_t}}


%%%%%%%%%%%%%%%%  intdifaut           %%%%%%%%%%

\def\rG{\mathrm{G}}
\def\brG{\overline{\rG}}
\def\Supp{{\rm Supp}}



%%%%%%%%%%%%%%  intdifline.tex     %%%%%%%%%%%%%%%%%%%%

\def\lind{{\rm l.ind}}
\def\rind{{\rm r.ind}}
\def\hmI1{\widehat{\mI_1}}
\def\tmI1{\widetilde{\mI_1}}
\def\tmJ1{\widetilde{\mJ_1}}
\def\hB1{\widehat{B_1}}
\def\hCB1{\widehat{\CB_1}}
\def\codim{{\rm codim}}
\def\bCF{\overline{\CF}}
\def\ffrac{{\rm frac}}
\def\bS{\overline{S}}


\def\linv{{\rm l.inv}}
\def\rinv{{\rm r.inv}}

%%%%%%%%%%%%%%  (larglquot.tex     %%%%%%%%%%%%%%%%%%%%

\def\Den{{\rm Den}}
\def\Denl{{\rm Den}_l}
\def\Ore{{\rm Ore}}
\def\ore{{\rm ore}}
\def\Den{{\rm Den}}
\def\LDen{{\rm LDen}}
\def\minLDen{{\rm min.LDen}}
\def\Loc{{\rm Loc}}
\def\minLoc{{\rm min.Loc}}
\def\Ass{{\rm Ass}}
\def\ttS{\widetilde{S}}
\def\ot{\overline{t}}
\def\maxDen{{\rm max.Den}}
\def\maxAss{{\rm max.Ass}}
\def\maxLoc{{\rm max.Loc}}
\def\llrad{{\rm l.lrad}}
\def\mW{\mathrm{W}}
\def\assmaxDen{{\rm ass.max.Den}}




%%%%%%%%%% genGoldie.tex   %%%%%%%%%%%%%%%

\def\OCC{\overline{\CC}}
\def\br{\overline{r}}
\def\bc{\overline{c}}
\def\bn{\overline{n}}
\def\gt{\mathfrak{t}}
\def\bs{\overline{s}}
\def\bt{\overline{t}}
\def\ga{\mathfrak{a}}

\def\tor{{\rm tor}}
\def\udim{{\rm udim}}

%%%%%%%%%%%%%%%%   locmaxrings.tex   %%%%%%%%%%%%%%%%

\def\bE{\overline{E}}
\def\Id{{\rm Id}}
\def\bpi{\overline{\pi}}
\def\Nil{{\rm Nil}}
\def\gll{\mathfrak{l}}
\def\oG{\overline{G}}


%%%%%%%%%%%%%%%  Crit-S-Simp-lQuot.tex   %%%%%%%%%%%%%%%%

\def\Mod{{\rm Mod}}
\def\RMod{$R$-{\rm Mod}}



%%%%%%%%%%%%%%   Crit-lNoeth-lQuot.tex  %%%%%%%%%%%%%%
\def\tCC{\widetilde{\mathcal{C}}}
\def\dCC{\mathcal{C}^\dagger}
\def\tQ{\widetilde{Q}}
\def\ts{\widetilde{\sigma}}
%\def\bI{\overline{I}}
\def\grR{{\rm gr}\, R}
\def\grQ{{\rm gr}\, Q}


%%%%%%%%%%%%%%%%%%%%%%%%%%  LocArtRing.tex   %%%%%%%%%%%%%%%%%

\def\IDen{{\rm IDen}}
\def\ILoc{{\rm ILoc}}
\def\maxker{{\rm max.ker}}
\def\CII{{\cal II}}
\def\CIT{{\cal IT}}



%%%%%%%%%%%%%%%%%%%%%%%%%%    Clas-lreg-quot.tex   %%%%%%%%%%%%%%%%%

\def\pCC{{}'\CC}
\def\pCCR{{}'\CC_R}
\def\pCCU{{}'\CC_U}
\def\pCCI{{}'\CC_I}
\def\pCCRJ{{}'\CC (R, J)}
\def\RR{{}_RR}
\def\pQ{{}'Q}
\def\pbQ{{}'\overline{Q}}
\def\pbCC{{}'\overline{\CC}}
\def\pCM{{}'\CM}
\def\bCM{\overline{\CM}}
\def\pSlR{{}'S_l(R)}
\def\pQlR{{}'Q_l(R)}
\def\pS{{}'S}
\def\pga{{}'\ga}
\def\pAsslR{{}'{\rm Ass}_l(R)}
\def\pD{{}'\D}
\def\pG{{}'\G}

%%%%%%%%%%%%%%%%%%%%    LocRingSets   %%%%%%%%%%%%%%%%%%%%%%%

\def\RSm1{R\langle S^{-1}\rangle}
\def\mLR{\mL (R)}

%%%%%%%%%%%%%%%%%%%%    SkewCatAlg   %%%%%%%%%%%%%%%%%%%%%%%

\def\ObCC{{\rm Ob} (\CC )}
\def\CCs{\CC (\s )}
\def\bCCs{\overline{\CC (\s )}}
\def\MorCC{{\rm Mor}(\CC )}
\def\CCsij{\CCs_{ij}}
\def\CCsii{\CCs_{ii}}
\def\CCsjj{\CCs_{jj}}
\def\bCCsij{\bCCs_{ij}}
\def\bCCsii{\bCCs_{ii}}
\def\bCCsjj{\bCCs_{jj}}
\def\bD{\overline{D}}
\def\bDhf{\overline{D}_{h(f)}}
\def\Ob{{\rm Ob}}
\def\rk{{\rm rk}}
\def\Irr{{\rm Irr}}
\def\lc{{\rm lc}}

%%%%%%%%%%%%%%%%%%  bav-lu  %%%%%%%%%%%%%%%%%
\def\sl2{\mathfrak{sl}_2}
\def\gl2{\mathfrak{gl}_2}
\def\gs{\mathbb{s}}
\def\gs{\mathfrak{s}}
\def\mK{\mathbb{K}}
\def\Prim{{\rm Prim}}
\def\Wt{{\rm Wt}}
\def\T{\mathbb{T}}
\def\msA{\mathscr{A}}
\def\sC{\mathscr{C}}
\def\Soc{{\rm Soc}}
\def\uE{\EuScript{E}}
\def\uH{\EuScript{H}}
\def\uX{\EuScript{X}}
\def\CW{\mathcal{W}}
\def\sX{\mathscr{X}}

% % % % % % % % % % % % % % % % % % % % % % % % % % % % % %
\def\Hlp{H_{\l}'}
\def\Elp{E_{\l}'}
\def\Flp{F_{\l}'}
\def\Dlp{\D_{\l}'}
\def\Clmn{C_{\l}^{\mu,\nu}}
\def\almn{{\sf a}_{\l}^{\mu, \nu}}
\def\sl2{\mathfrak{sl}_2}
\def\gs{\mathbb{s}}
\def\gs{\mathfrak{s}}
\def\mK{\mathbb{K}}
\def\Prim{{\rm Prim}}
\def\Wt{{\rm Wt}}
\def\T{\mathbb{T}}
\def\msA{\mathscr{A}}
\def\sC{\mathscr{C}}
\def\Soc{{\rm Soc}}
\def\CW{\mathcal{W}}
\def\CO{\mathcal{O}}
\def\Irr{{\rm Irr}}

\def\Am{A(m)}
\def\CDAm{{\cal D} (A(m))}


\begin{document}
\begin{thebibliography}{3}
\bibitem{Aus} M. Auslander, \emph{On the dimension of modules and algebras (3), Global dimension}, Nagoya Math. J. {\bf 9} 1955, 67--77.

\bibitem{Bav-GWA-FA-91} V. V. Bavula,  \emph{Finite-dimensionality of ${\rm Ext}\sp n$ and ${\rm Tor}\sb n$ of simple modules over a class of algebras}, Funct. Anal. Appl., No. \textbf{3}, {\bf 25}, 1991, 229--230.

\bibitem{Bav-GWA-Clas-sl2-90} V. V. Bavula, \emph{Classification of simple ${\rm sl}(2)$-modules and the
finite-dimensionality of the module of extensions of simple ${\rm sl}(2)$-modules}, Ukrainian Math. J. {\bf 42}, No. \textbf{9}, 1990,  1044--1049.

\bibitem{Bav-UkrMathJ-92} V. V. Bavula, \emph{Simple $D[X,Y;\sigma,a]$-modules}, Ukrainian Math. J., {\bf 44} No. 12, 1992, 1500--1511.

\bibitem{Bav-GWArep} V. V. Bavula, \emph{Generalized Weyl algebras and their representations (Russian)}, Algebra i Analiz {\bf  4}, No. 1, 1992, 75--97; translation in St. Petersburg Math. J. {\bf 4}, No. 1, 1993, 71--92.

\bibitem{Bav-GWAKernel-1993}  V. V. Bavula, \emph{Generalized Weyl algebras, kernel and tensor-simple algebras, their simple modules. {\it Representations of algebras}} (Ottawa, ON, 1992), 83--107, CMS Conf. Proc., 14, Amer. Math. Soc., Providence, RI, 1993.

\bibitem{Bav-Bav-ExtrMod-93} V. V. Bavula,  \emph{Extreme modules over the Weyl algebra $A\sb n$}, Ukrainian Math. J., {\bf 45} No. {\bf 9}, 1993, 1327--1338.

\bibitem{Bav-Bav-Ideals-I-93} V. V. Bavula, (1993) \emph{Description of two-sided ideals in a class of noncommutative rings. I.}, Ukrainian Math. J. {\bf 45}, No. {\bf 2}, 1993, 223--234.

\bibitem{Bav-Bav-Ideals-II-93} V. V. Bavula, \emph{Description of two-sided ideals in a class of noncommutative rings II.}, Ukrainian Math. J., {\bf 45} No. 3, 1993, 329--334.

\bibitem{Bav-FilDimSimCrit-1996} V. V. Bavula, \emph{Filter dimension of algebras and modules, a simplicity
criterion of   generalized Weyl algebras}, Comm. Algebra,  {\bf 24}, No. 6, 1996, 1971--1992.

\bibitem{Bav-GlGWA-1996} V. V. Bavula, \emph{Global dimension of generalized Weyl algebras}, Representation theory of algebras, (Cocoyoc, 1994), 81--107, CMS Conf. Proc., 18, Amer. Math. Soc.,   Providence,  RI, 1996.

\bibitem{Bav-1996THM} V. V. Bavula, \emph{Tensor homological minimal algebras, global dimension of the tensor product of algebras and of generalized Weyl algebras}, Bull. Sci. Math., {\bf 120}, 1996, 293-335.

%\bibitem{Bav-2016-gwa-diskew} V. V. Bavula,  Generalized Weyl algebras and diskew polynomial rings. Arxiv:1612.08941.

\bibitem{Bav-2017QGWA} V. V. Bavula, \emph{Quiver Generalized Weyl algebras, skew category algebras and diskew polynomial rings}, Mathematics in Computer Science, 2017,  DOI: 10.1007/s11786-017-0313-5, published online on 28 April 2017.

\bibitem{Bav-Oys-Kdim} V. Bavula and F. van Oystaeyen, \emph{Krull Dimension of Generalized Weyl Algebras and Iterated Skew Polynomial Rings: Commutative Coefficients},  J. of Algebra, {\bf 208}, No. {\bf 1}, 1998, 1-34.

\bibitem{Benkart-Roby-1998} G. Benkart, T. Roby, \emph{Down-up algebras}, J. Algebra, {\bf  209}, 1998, 305--344.

\bibitem{Cassidy-2003} T. Cassidy, \emph{Homogenized down-up algebras}, Comm. Algebra, {\bf 31 (4)}, 2003, 1765--1775.

\bibitem{Kirk-Kuzm-2005} E. E. Kirkman and J. Kuzmanovich, \emph{Fixed subrings of Noetherian graded regular rings}, J. of Algebra, {\bf 288}, 2005, 463--484.

 \bibitem{Kirk-Mus-Pas-1999} E. Kirkman, I. Musson, D. Passman, \emph{Noetherian down-up algebras}, Proc. Amer. Math. Soc., {\bf  127}, 1999, 3161--3167.

\bibitem{KS} E. E. Kirkman and L. W. Small, \emph{q-analogs of harmonic oscillators and related rings, preprint}, Israel J. Math., {\bf 81}, 1993, 111--127.

\bibitem{Ma} M.-P. Malliavin, \emph{L'alg\`ebre d'Heisenberg quantique}, C. R. Acad. Sci. Paris, S\'er. 1, {\bf 317}, 1993, 1099-1102.

\bibitem{MR} J. C. McConnell and J. C. Robson, \emph{Noncommutative Noetherian rings}, Wiley, Chichester,  1987.

\bibitem{Musson-1986} I. M. Musson,  \emph{Some rings of differential operators which are Morita equivalent to the Weyl algebra $A_1$}, Proc. Amer. Math. Soc., {\bf 98}, No. {\bf 1}, 1986, 29--30.

\bibitem{Smith-SimUsl2-1990} S. P. Smith,  \emph{A class of algebras similar to the enveloping algebra of $sl(2)$}. Trans. Amer. Math. Soc.,  {\bf 322}, No. {\bf 1}, 1990, 285--314.

\bibitem{Smith1} S. P. Smith,  \emph{Quantum qroups: An introduction and survey for ring theorists, in Noncommutative Rings (S.Montgomery and L.W.Small, Eds.)}, pp. 131--178, MSRI publ. 24, Springer-Verlag, Berlin 1992.

%\bibitem{Smith-Staff} S. P. Smith, and J. T. Stafford,
%Differential operators on an affine curve. {\it Proc. London Math. Soc.}  (3) {\bf 56} (1988), no. 2, 229â€“-259.

\bibitem{Za} C. Zachos, \emph{Elementary paradigms of quantum algebras}, Contemporary Math., {\bf 134}, 1992, 351-377.

\bibitem{Ros} A. L. Rosenberg, \emph{The spectrum of the algebra of skew differential operators and the irreducible representations of the quantum Heisenberg algebra}, Comm. Math. Phys. {\bf 142}, 1991, 567-588.

\end{thebibliography}


\title{Rings of differential operators on singular generalized multi-cusp algebras}%Paper title
\author{V. V. \  Bavula  and K. Hakami}%Authors names
\shorttitle{}%Short paper title for the headers
\shortauthor{V. V. \  Bavula, K. Hakami}%Short authors names for the headers
\address[V. V. \  Bavula]{Department of Pure Mathematics, University of Sheffield, Hicks Building, Sheffield S3 7RH, UK}{v.bavula@sheffield.ac.uk}{}%www.fauthor.domain.com
\address[Khalil Hakami]{Department of Mathematics, College of Science, Jazan University, P.O. Box. 114, Jazan 45142, Kingdom of Saudi Arabia}{khakami@jazanu.edu.sa}{www.sauthor.domain.com}

\thanks{Type your thanks here..}

\communicated{Communicated person}

\research% Type of the paper. \survey command may be used also.

\date{11.05.2003}

%Change theorem environments according to your needs...
\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}{Lemma}
\newtheorem{proposition}{Proposition}
\newtheorem{corollary}{Corollary}
\newtheorem{definition}{Definition}
\theoremstyle{definition}
\newtheorem{example}{Example}
\newtheorem{remark}{Remark}

\maketitle

\begin{abstract}
The aim of the paper is to study the ring of differential operators $\CDAm$ on the generalized multi-cusp algebra $A(m)$ where $m\in \N^n$ (of Krull dimension $n$). The algebra $\Am$ is singular apart from the single case when $m=(1, \ldots , 1)$. In this case, the algebra $\Am$ is a polynomial algebra in $n$ variables. So, the $n$'th Weyl algebra $A_n=\CD (A(1, \ldots , 1))$ is a member of the family of algebras $\CDAm$.
We prove that the algebra $\CDAm$  is a central, simple, $\Z^n$-graded,  finitely generated Noetherian domain of Gelfand-Kirillov dimension $2n$.
 Explicit finite sets of generators and defining relations is given for the algebra $\CDAm$. We prove that the Krull dimension and the global dimension of the algebra $\CDAm$ is $n$. An analogue of the Inequality of Bernstein is proven. In the case when $n=1$, simple $\CD (\Am)$-modules are classified.
\end{abstract}

\subjclass{2001}{16S32, 16E10, 16E05, 16D60,  16P40,  16S35, 16S15, 16P50, 16P90, 16D30.}

\keywords{the ring of differential operators, the generalized  multi-cusp algebra,  the generalized Weyl algebra, the global dimension, the Krull dimension, the Gelfand-Kirillov dimension, simple module, the projective dimension,  orbit,  the projective resolution.}

\section*{Introduction}

The following notation will remain  fixed throughout the paper (if
it is not stated otherwise): $K$ is a field of  characteristic zero
(not necessarily algebraically closed), module means a left
module, $P_n=K[x_1, \ldots , x_n]$ is a polynomial algebra over
$K$, $\der_1:=\frac{\der}{\der x_1}, \ldots ,
\der_n:=\frac{\der}{\der x_n}\in \Der_K(P_n)$,
$A_n=K\langle x_1, \ldots , x_n, \der_1, \ldots , \der_n \rangle \subseteq \End_K(P_n)$ is the $n$'th {\em Weyl algebra} ($A_n=\CD (P_n)$ is the algebra of differential operators on the polynomial algebra $P_n$),
 $\N =\{ 0, 1, \ldots \}$ is the set of natural numbers and  $\N_{\geq s}=\{ i\in \N\, | \, i\geq s\}$. In the  case $n=1$, we usually drop the subscript `1'. So, $P=K[x]$ is a polynomial algebra in a variable $x$,
$A_1=K \langle x , \der \; | \;  \der x - x \der = 1 \rangle $
is the {\em Weyl algebra}, i.e., $A_1=\CD (P)$
is the ring of differential operators on the polynomial algebra $P$.

The algebra of regular functions on the cusp $y^2=x^3$ is isomorphic to the subalgebra $A(2)=K + \sum_{i \geq 2} K x^{i}$ of the polynomial algebra $P=K[x]$. For each $m \geq 1$,
$A=A(m)=K + \sum_{i \geq m} K x^i$ is a subalgebra of $P$ which is caleld the {\bf generalized cusp algebra}. Clearly, $A(1)=K[x]$ is a polynomial algebra and $A(2)$ is the cusp algebra. \\

 {\bf Definition.}  Let $m=(m_1, \ldots ,m_n)\in \N^n$, the subalgebra  of the polynomial algebra $P_n=K[x_1, \ldots , x_n]$,
 $$\Am= \bigotimes_{i=1}^n A(m_i), \;\; {\rm where}\;\;  A(m_i) = K + \sum_{j \geq m_i} K x_i^j\subseteq K[x_i],$$ is called the {\bf generalized multi-cusp algebra} of rank $n$ (GMCA, for short).\\

Clearly, if $m=(1, \ldots , 1)$ then $\CD(A(m)) = A_n$ is the $n$'th Weyl algebra.
 If if $m=(2, \ldots , 2)$ then $A(m)\simeq A(2)^{\t n}$ is the algebra of regular functions on the direct product of $n$ copies of the cusp.

 The aim of the paper is to study algebraic properties of the algebra $\CDAm$
of differential operators of the generalized multi-cusp algebra $\Am$ of rank $n$.  We are mostly interested in the case when $m=(m_1, \ldots , m_n)\in \N_{\geq 2}^n$ since for an arbitrary $m$ the algebra $\Am$ is isomorphic to the tensor product $P_s\t A(m')$ where $m'\in \N_{\geq 2}^{n-s}$ and $\CDAm\simeq A_s\t \CD (A(m'))$. \\


{\bf Generators and defining relations for the algebra $\CDAm$.} Theorem \ref{GenRel30Jan20} gives an explicit finite sets of generators and defining relations of the algebra $\CDAm$.



\begin{theorem}\label{GenRel30Jan20}%\marginpar{GenRel30Jan20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$.  Then
\begin{enumerate}
\item $\CDAm\simeq \bigoplus_{i=1}^n\CD (A(m_i))$.
\item For each $i=1, \ldots , n$, let $\CG_i$ and $\CR_i$ be the set of generators and defining relations of the algebra $\CD (A(m_i))$ as in  Theorem \ref{AB18Nov18}.(4). Then the algebra $\CDAm$ is generated by the finite set of elements $\CG = \cup_{i=1}^n\CG_i$ that satisfy the defining relations $\CR_1, \ldots , \CR_n$ and $g_ig_j=g_jg_i$ for all $g_i\in \CG_i$ and $g_j\in \CG_j$ for all $i\neq j$.
\end{enumerate}
\end{theorem}

 A $K$-algebra $R$ is called {\em central} if its centre $Z(R)$ is equal to the field $K$. Theorem \ref{GenProp30Jan20} is about general properties of the algebra $\CDAm$.
\begin{theorem}\label{GenProp30Jan20}%\marginpar{GenProp30Jan20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$.  Then The algebra $\CDAm$ is a central, simple, $\Z^n$-graded,  finitely generated Noetherian domain of Gelfand-Kirillov dimension $2n$.
\end{theorem}


{\bf An analogue of the Inequality of Bernstein for the algebras  $\CD (\Am )$.} The starting point of the $\CD$-module theory is the {\em Inequality of Bernstein}: {\em For  all nonzero finitely generated $A_n$-modules} $M$, $\GK (M)\geq n$.

\begin{theorem}\label{BerIn30JAn20}%\marginpar{BerIn30JAn20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$. For  all nonzero finitely generated $\CDAm$-modules $M$, $\GK (M)\geq n$.
\end{theorem}




{\bf The Krull and global dimensions of the algebra $\CD (\Am )$.} The Krull dimension of the Weyl algebra $A_n$ is $n$, \cite{MR}.

\begin{theorem}\label{Kdim30JAn20}%\marginpar{Kdim30JAn20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$. The Krull  dimension of the algebra $\CDAm $ is $n$.
\end{theorem}
 The global  dimension of the Weyl algebra $A_n$ is $n$, \cite{MR}.


\begin{theorem}\label{gldim30JAn20}%\marginpar{gldim30JAn20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$. The global dimension of the algebra $\CDAm $ is $n$.
\end{theorem}

{\bf Classification of simple $\CD (A)$-modules where $A=A(m)=K+\sum_{i\geq m}Kx^i$.}

%*** Block, GWA, others ****

The set $\widehat{\CD (A)}$ of isomorphism classes of simple $\CD (A)$-modules is a disjoint union of two subsets: the set of $D$-torsion and the set of $D$-torsion free simple $\CD (A)$-modules where $D=K[h]$ and $h=\der x$. The sets of simple $D$-torsion and  $D$-torsion free  $\CD (A)$-modules are classified in Theorem \ref{DTor27Jan20} and Theorem \ref{DTFr27Jan20}, respectively.

















%%%%%%%%%%%%%%%%%% SECTION 2 %%%%%%%%%%%%%%%%%%%%%%%%



\section{Generators and defining relations of the algebra $\CD (A)$}\label{PRLMN}%\marginpar{PRLMN}

The aim of this section is to find generators and defining relatyions of the algebra $\CD (A)$ of differential operators on the algebra $A=\Am$ (Theorem \ref{AB17Nov18}). It is proven that the algebra $\CD (A)$ is a central simple Noetherian domain of Gelfand-Kirillov dimension 2 (Theorem \ref{AB18Nov18}.(1)).
The Krull dimension of the algebra $\CD (A)$ is 1 (Theorem \ref {E29Jan20}).
 Furthermore, for all  nonzero left ideals $I$  of the algebra $\CD (A)$,  the $\CD (A)$-module $\CD (A)/I$ has finite length (Theorem  \ref{C29Jan20}). We introduce two generalized Weyl algebras $\mA$ and $\mB$ such that $\mA\subset \CD (A)\subset \mB = T^{-1}\mA \simeq  T^{-1}\CD (A)$.
 The properties of the algebra $\CD (A)$ is a mixture of properties of the algebras $\mA$ and $\mB$. \\








{\bf Generalized Weyl algebras $D(\s , a)$ of rank 1,  \cite{Bav-GWA-FA-91}-\cite{Bav-Bav-Ideals-II-93}}. Let $D$ be a ring, $\s $ be a ring automorphism of $D$,
 $a$ is a {\em central} element of $D$.
 The {\bf generalized Weyl algebra} of rank 1 (GWA, for short)
 $D(\s , a)=D[X,Y; \s , a]$ is a ring
 generated by the ring $D$  and two elements $X$ and $Y$
 that are subject to the defining relations:
%\marginpar{clGWA}
\begin{equation}\label{clGWA}
 Xd=\s (d) X\;\;  {\rm and} \;\;  Yd=\s^{-1} (d)Y\;\; {\rm for \; all} \;\; d\in D,  \;\;
 YX=a \;\; {\rm and} \;\;   XY=\s (a).
\end{equation}
The ring $D$ is called the {\em base ring} of the GWA.
The automorphism $\s$ and the element $a$ are
called the {\em defining automorphism} and
the {\em defining element} of the GWA, respectively.


The algebra $A=\bigoplus_{n\in {\bf \Z}}\, A_n$
is $\Z$-graded where $A_n=Dv_n$,
$v_n=X^n$ and  $v_{-n}=Y^n$ for $n<0$,  and $v_0=1.$
 It follows from the above relations that
$v_nv_m=(n,m)v_{n+m}=v_{n+m}\langle n,m\rangle $
for some $(n,m)\in D$. If $n>0$ and $m>0$ then
\begin{eqnarray*}
 n\geq m:& &  (n,-m)=\sigma^n(a)\cdots \sigma^{n-m+1}(a),\;\;  (-n,m)=\sigma^{-n+1}(a)
\cdots \sigma^{-n+m}(a),\\
n\leq m: & & (n,-m)=\sigma^{n}(a)\cdots \sigma(a),\,\,\,\;\;\; \;\;\; \;\;\; (-n,m)=\sigma^{-n+1}(a)\cdots a,
\end{eqnarray*}
in other cases $(n,m)=1$. Clearly, $\langle n,m\rangle =\s^{-n-m}((n,m))$.\\


{\bf Example.} The Weyl algebra $A_1=K[h][x, \der ; \s , a=h]$ is a GWA where $h=\der x$ and  $\s (h)=h-1$. \\


{\bf Generators and defining relations of the algebra $\CD (A)$.}
 The set $S_{x} = \{ x^{i} \; | \;  i \geq 0 \}$
(resp., $S_{x^m} = \{ x^{i  m} \; | \;  i \geq 0 \}$)
is a multiplicative set of $P$
(resp., $P$ and $A$).
Clearly,
%\marginpar{SxmPA}
\begin{equation}\label{SxmPA}
K[x,x^{-1}] = S_{x}^{-1} P = S_{x^m}^{-1} P = S_{x^m}^{-1} A.
\end{equation}

The polynomial algebra $P$
is a left $A_1$-module which is isomorphic to
the factor module $A_1 / A_1 \der$
where the action of $A_1$ is given by the rule:
For all $p \in P$, $x \ast p = xp$ and $\der \ast p = p := \frac{dp}{dx}$.
The left $A_1$-module $P=\bigoplus_{i \geq 0} K x^{i}$ is
a $\Z$-graded
(even $\N$-graded) $A_1$-module  and
$h \ast x^{i} = (i +1) x^{i} \;\; {\rm  for \; all} \;\;  i \geq0 \;\; {\rm  where} \;\; h= \der x $.

\begin{theorem}\label{AB17Nov18}%\marginpar{AB17Nov18}
Let $K$ be a field of characteristic zero,
$A=K + \sum_{i \geq m} K x^{i}$ \; $(m \geq 2)$ be
a subalgebra of the  polynomial algebra $P=K[x]$.
Then
\begin{enumerate}
\item The ring of differential operators $\CD (A)$ on $A$
is a $\Z$-graded subalgebra
$ \CD (A) = \bigoplus_{i \in \Z} \CD (A)_{[i]}$
of the $\Z$-graded algebra $A_{1,x}$ where $\CD (A)_{[i]} = D \d_i$ and
\[
\d_i = \begin{cases}
            x^{i} & \text{if} \, \, \,  i \geq m,  \\
             (h-i-1) x^{i}  & \text{if} \, \, \,  i=1, \ldots , m-1, \text{and} \\
             1  & \text{if} \, \, \, i=0, \\
             \end{cases}
\]
\[
\d_{-i} = \begin{cases}
            (h+i-1) \cdot \prod_{j=m-i+1}^m (h-j) x^{-i} & \text{if} \, \, \,  i = 1, \ldots, m-1,   \\
             (h+i-1) \cdot \prod_{1\neq j=m-i + 1}^m (h-j) x^{-i}  & \text{if} \, \, \, i \geq m. \\
             \end{cases}
\]
In particular, $\d_{-m}=(h+m-1) (h-2) \cdots (h-m) x^{-m}$, and for all $i\in \Z$, $\d_i=\v_ix^i$ where the polynomial $\v_i\in D=K[h]$ is  the coefficient of $x^i$ in the equalities above.
\item For all $i,j \geq m$, $\d_{-i} \d_{-j} = \d_{-i-j}$ and $\d_i\d_j=\d_{i+j}$.
\item   $ \CD(A) = \bigoplus_{j\geq 0, m\leq i\leq 2m-1} D\d_{-i}\d_{-m}^j\oplus \bigoplus_{|i|<m}D\d_i\oplus \bigoplus_{j\geq 0, m\leq i\leq 2m-1} D\d_i\d_m^j $, and  $\d_{-i}\d_{-m}^j=\d_{-m}^j\d_{-i}$ and $\d_i \d_m^j=\d_m^j\d_i $ for all $j\geq 0$ and $ m\leq i\leq 2m-1$.

\item The algebra $\CD(A)$ is  generated algebra by the elements
$ \{ h , \d_i \; | \;  i=\pm 1, \pm 2, \ldots, \pm (2m-1) \}$ that satisfy the finite set of defining relations: For all $i,j=\pm 1, \ldots , \pm (2m-1)$,  $[h, \d_i]=i\d_i$  and
$$
\d_i\d_j=\begin{cases}
\v_i\s^i (\v_j)\v_{i+j}^{-1} \d_{i+j}& \text{if } |i+j|<2m,\\
\v_i\s^i (\v_j)\v_{i+j-m}^{-1}\d_{i+j-m}\d_m & \text{if } 2m\leq i+j<3m,\\
\v_i\s^i (\v_j)\v_{i+j-2m}^{-1}\d_{i+j-2m}\d_m^2 & \text{if } 3m\leq i+j<4m,\\
\v_i\s^i (\v_j)\v_{i+j+m}^{-1}\d_{i+j+m}\d_{-m} & \text{if } -3m< i+j\leq -2m,\\
\v_i\s^i (\v_j)\v_{i+j+2m}^{-1}\d_{i+j+2m}\d_{-m}^2 & \text{if } -4m< i+j\leq -3m. \\
\end{cases}
$$
\end{enumerate}
\end{theorem}

{\it Proof}. 1. The set $S_x =\{ x^{i} \; | \;  i \geq 0 \}$
(resp., $S_{x^m} = \{ x^{mi} \; | \;   i \geq 0 \}$)
is an Ore set of the Weyl algebra $A_1$
(resp., of $A_1$ and $\CD (A)$)
and
%\marginpar{ASAD}
\begin{equation}\label{ASAD}
A_{1,x} := S_{x}^{-1} A_1 = S_{x^m}^{-1} A_1 = S_{x^m}^{-1} \CD(P) \simeq  \CD (S_{x^m}^{-1} P) \stackrel{(\ref{SxmPA})} = \CD (S_{x^m}^{-1} A) \simeq  S_{x^m}^{-1} \CD (A).
\end{equation}
Recall that the Weyl algebra
$A_1 = D [x, \der ; \s , a = h ]$ is GWA
when $D = K[h]$, $\s (h) = h-1$ and $h:= \der x$.
In particular, the Weyl algebra
$A_1 = \bigoplus_{i \in \Z} D v_i$ is a $\Z$-graded algebra
where $v_{0} := 1$, $v_{i} = x^{i}$ and $v_{-i} = \der^{i}$  for $i \geq 1$.

Since the elements of the Ore set $S_x$
%and $S_{x^m}$
are homogeneous elements of the algebra $A_1$,
the localized algebra $A_{1,x} =S_x^{-1} A_1$ is also a $\Z$-graded algebra
$A_{1,x} = \bigoplus_{i \in \Z} D x^i$ (since $\der = h x^{-1}$).
By (\ref{ASAD}),
$\CD (A) = \{ \d \in A_{1,x} \; | \; \d \ast A \subseteq A  \}$.

Since the algebra $A$ is a $\Z$-graded subalgebra of the polynomial algebra $P$, the algebra $\CD (A)$ is also $\Z$-graded,
%\marginpar{ASAD1}
\begin{equation}\label{ASAD1}
\CD (A) = \bigoplus_{i \in \Z} \CD (A)_{[i]}  \;\; {\rm  where} \;\;  \CD(A)_{[i]} = \CD(A) \cap D x^i = \{ \d \in D x^i  \; | \;  \d \ast A \subseteq A \}.
\end{equation}

Now, using the fact that $ h \ast x^i = (i+1) x^i$
for all $i \in \Z$,
we obtain the explicit expressions for the graded components
$\CD (A)_{[i]}$ as in the theorem.

2.  Statement 2 follows at once from the definition of the elements
$\d_{-i}$ and $\d_i=x^i$ $(i \geq m)$  and the fact that
$x^{-j} h = (h +j) x^{-j} $ for all $j \geq 0$.

3.  By statement 2, for all $i \geq 1$ and $j= 0,1, \ldots, m-1$,
$\d_{-im -j} = \d_{-m}^{i} \d_{j}$ and $x^{im+j}=(x^m)^i \cdot x^j$.
Now, statement 3 follows from statement 1.


4. By statements 1 and 2, the relations in statement 4 hold.
 Then the  relations of statement 4 are defining relations of the algebra $\CD (A)$ since they imply  the first equality in statement 3 where the direct sums are replaced by sums. $\Box$\\

{\bf The subalgebra $\CA$ of $\CD (A)$ which is a simple GWA.}
 For an automorphism $\tau$ of an algebra $R$, $R^\tau := \{ r\in R\, | \, \tau (r)=r\}$ is the {\em algebra of $\tau$-constants/invariants}. The subalgebra $\CA$ of $\CD (A)$ (Proposition \ref{AB18Nov18}.(2)) plays a key role in proving that the algebra $\CD (A)$ is a simple Noetherian domain (Proposition \ref{AB18Nov18}.(1)).

\begin{theorem}\label{AB18Nov18}%\marginpar{AB18Nov18}
Let $K$ be a field of characteristic zero,
$A=K + \sum_{i \geq m} K x^{i} \;\; (m \geq 2)$
be a subalgebra of the  polynomial algebra $P=K [x]$.  Then
\begin{enumerate}
\item The algebra $\CD (A)$ is a central simple  Noetherian domain.
\item The subalgebra $\CA$ of $\CD (A)$ which is generated by the elements
$h$, $X:= x^m$ and $Y:= \d_{-m}$ is a  GWA
 $\CA = D[X,Y;\s^m, a=(h+m-1) \cdot (h-2) (h-3) \cdots (h-m)]$ which  is a central simple  Noetherian domain
where $\s (h) = h-1$.
\item The algebra $\CD(A)$ is a finitely generated  left and right $\CA$-module,
$ \CD(A) = \sum_{|i| < 2m} \CA \d_i = \sum_{|i| < 2m} \d_{i} \CA $.
\end{enumerate}
\end{theorem}

{\it Proof}.  $2$.  The elements $h$, $X=x^m$ and $Y=\d_{-m}$ satisfy
the defining relations for the GWA
$D[X , Y ; \s^m , a]$.
Then using the fact that the algebra $\CA$ is
a homogeneous subalgebra of the $\Z$-graded algebra $\CD (A)$,
we see that
$\CA = \bigoplus_{i \geq 1} D Y^i \oplus D \oplus \bigoplus_{i \geq 1} D X^i$, and so $\CA = D [X , Y, \s^m , a]$ since $a=YX= (h+m-1) \cdot (h-2) (h-3) \cdots (h-m)x^{-m}x^m=(h+m-1) \cdot (h-2) (h-3) \cdots (h-m)$, by Theorem \ref{AB17Nov18}.(1). By  \cite[Corollary 3.2]{Bav-GWArep}, the GWA $\CA$  is a simple algebra since the difference of any two distinct roots of the polynomial
 $a=(h+m-1) \cdot (h-2) (h-3) \cdots (h-m)$ is not divisible by $m$.
 By  \cite[Proposition 1.3]{Bav-GWArep}, the GWA $\CA$  is a noetherian domain. Clearly, $Z(\CA ) = D^\s  = K$ since $\s (h)=h-1$ and the field $K$ has characteristic zero.


$3$.  By using the definition,
the algebra $\CA$ is generated by the elements
$h$, $x^m$ and $\d_{-m}$.
Now, statement 3 follows from Theorem \ref{AB17Nov18}.(2,3).

$1$.   (i) {\em The algebra $\CD (A)$ is a Noetherian domain}:
Since the polynomial algebra $D=K[h]$ is a Noetherian algebra,
the GWA $\CA$ is also a  Noetherian domain \cite[Proposition 1.3]{Bav-GWArep}.
The algebra $\CD(A)$ is a  finitely generated left and right $\CA$-module.
Hence, the algebra $\CD(A)$ is a Noetherian left and right $\CA$-module.
Therefore, the algebra $\CD(A)$ is a Noetherian algebra.

 (ii) {\em The algebra $\CD (A)$ is simple}: Let $I$ be a nonzero ideal of the algebra $\CD (A)$. Then $\ga := I\cap D\neq 0$ is a nonzero ideal of the algebra $D$ since the algebra $\CD (A)$ is a domain which is a direct sum $\CD (A)=\bigoplus_{i\in \Z} D\d_i$ of eigen-spaces of the inner derivation $\ad_h $ of the algebra $\CD (A)$ (Theorem \ref{AB17Nov18}.(1)). The subalgebra $\CA$ of $\CD (A)$ is a simple algebra (statement 2) that contains the algebra $D$. Hence, $0\neq \ga \subseteq I\cap \CA $ is a nonzero ideal of the algebra $\CA$, i.e. $1\in I\cap \CA\subseteq I$,  and so  $I=\CD (A)$. Therefore, the algebra $\CD (A)$ is a simple algebra.

(iii) {\em The algebra $\CD (A)$ is central}: By (\ref{ASAD}), the algebra $\CD (A)$ is a central algebra:
$$K\subseteq Z(\CD (A))\subseteq Z(S_{x^m}^{-1}\CD (A))) =Z(A_{1,x}) =K.\;\; \Box$$

The set $S_{x^m}=\{ x^{im}\, | \, i\geq 0\}$ is a denominator set of the algebras $\CA$ and $A_1=\CD (P)$. The set $S_{x}=\{ x^{i}\, | \, i\geq 0\}$ is a denominator set of the Weyl  algebra $A_1=\CD (P)$. We have the following inclusions of algebras
%\marginpar{AAA1xm}
\begin{equation}\label{AAA1xm}
 A_1\subset A_{1, x^m}=A_{1, x}=D[x,x^{-1}; \s], \;\; D=K[h],  \;\; \s (h) = h-1,
\end{equation}
%\marginpar{AAA1xm1}
\begin{equation}\label{AAA1xm1}
 \CD (A )\subset \CD (A )_{ x^m}\simeq A_{1, x}=D[x,x^{-1}; \s],
\end{equation}
%\marginpar{AAA1xm2}
\begin{equation}\label{AAA1xm2}
\CA \subset \CA_{x^m}= D[x^m,x^{-m}; \s^m]\subset  A_{1, x^m}= A_{1, x}=D[x,x^{-1}; \s],
\end{equation}
where the subscripts `$x^m$' and `$x$' denote the (left and right) localizations at the denominator sets $S_{x_m}$ and $S_x$, respectively.
The rings  $D[x^{-1}, x; \s ]$ and $D[x^m,x^{-m}; \s^m]$ are  skew Laurent polynomial rings.


Recall that the Weyl algebra $\CD(P)=A_1$ is
the GWA,  $A_1=D[x, \der ; \s, h] = \bigoplus_{i \in \Z} D v_i$,
where $v_0=1$, $v_i = x^i$ and $v_{-i}= \der^{i}$ for all $i \geq 1$.
Since $\der x = h$,
we have that $ x^{-1} = h^{-1} \der$.
Then, for all $i \geq 1$,
%\marginpar{xmiph}
\begin{equation}\label{xmiph}
x^{-i} = \prod_{k=0}^{i-1} (h+k)^{-1} \der^i.
\end{equation}
Now, for $i=1$,
%\marginpar{xmiph1}
\begin{equation}\label{xmiph1}
\d_{-1} =\dfrac{h (h-m)}{h} \der = (h-m) \der.
\end{equation}

For $i=2, \ldots, m-1 $,
%\marginpar{xmiph2}
\begin{equation}\label{xmiph2}
\d_{-i} =\dfrac{(h+i-1)\prod_{j=m-i+1}^{m} (h-j)}{\prod_{k=0}^{i-1} (h+k) } \der^i
 = \dfrac{\prod_{j=m-i+1}^{m} (h-j)}{\prod_{k=0}^{i-2} (h+k) } \der^i .
\end{equation}

For $i \geq m $,
%\marginpar{xmiph3}
\begin{equation}\label{xmiph3}
\d_{-i} =\dfrac{(h+i-1) \prod_{1 \neq j = -i +m +1 }^{m} (h-j) }{ \prod_{k=0}^{i-1} (h+k) } \der^i
= \dfrac{\prod_{j=2}^{m} (h-j)}{ \prod_{k=i-m}^{i-2} (h+k) } \der^i .
\end{equation}

\begin{corollary}\label{aAB17Nov18}%\marginpar{aAB17Nov18}
Let $A$ be as in Theorem \ref{AB17Nov18}.
\begin{enumerate}
\item $\CD (A) \not \subseteq \CD (P)$.
\item Let $\d_{-i}$, $i \geq 1$ be as in Theorem \ref{AB17Nov18}.
Then $\d_{-1} = \CD (P)$ and $\d_{-i} \not \in \CD (P)$ for $i \geq 2$.
\end{enumerate}
\end{corollary}

{\it Proof}.
$1$.
Statement 1 follows  from statement 2.

$2$.
Statement 2 follows from (\ref{xmiph1}), (\ref{xmiph2}) and (\ref{xmiph3}). $ \Box$

In Theorem \ref{BB18Nov18}, the algebra
$\CD(A) \cap \CD(P)$ is described and an explicit set of algebra generators is given for it. \\


{\bf The subalgebra $\CD (A)_{+}$ and $\CD (A)_{-} $ of $\CD (A)$}.
Let the algebra $A$ be as in Theorem \ref{AB17Nov18}.
The algebra $\CD(A)$ contains two homogeneous subalgebras
$ \CD (A)_{+} := \bigoplus_{i \geq 0} D \d_i $ and
$ \CD(A)_{-} := \bigoplus_{i \geq 0} D \d_{-i} $.

\begin{proposition}\label{CB18Nov18}%\marginpar{CB18Nov18}
Let $A$ be as in Theorem \ref{AB17Nov18}.
\begin{enumerate}
\item The algebras  $\CD(A)_{\pm} $ are finitely generated Noetherian algebras.
\item $\CD (A)_{+} \subseteq \CD (P)$ but $\CD (A)_{-} \not \subseteq \CD (P)$.
\item The algebra $ \CD (A)_{+} $ is a finitely generated, left and right module
over its subalgebra $D [x^m ; \s^m]$ and
the set $ \{ 1, \d_1, \ldots, \d_{2m-1} \}$
is a module generating set.
\item The algebra $\CD (A)_{-}$ is a finitely generated, left and right module
over its subalgebra $D [\d_{-m} ; \s^{-m}]$ and
the set $ \{ 1, \d_{-1}, \ldots, \d_{-2m+1} \}$ is a module generating set.
\end{enumerate}
\end{proposition}

{\it Proof}.
$2$.
The inclusion $\CD (A)_{+} \subseteq \CD (P)$ is obvious.
By Corollary \ref{aAB17Nov18}.(2), $\CD(A)_{-} \not \subseteq \CD(P)$.

$3$.
Statement 3 follows at once from the explicit expressions for
the elements $\d_i  \;\; (i \geq 0)$ and the fact that $\CD(A)_{+} = \bigoplus_{i \geq 0} D \d_i$.

$4$.
Statement 4 follows from Theorem \ref{AB17Nov18}.(2,3).

$1$.
The skew polynomial rings $D[x^m ; \s^m]$ and $D[\d_{-m} ; \s^{-m}]$
are Noetherian algebras (since $D$ is so).
Now, statement 1 follows at once from statement 4. $\Box$\\


{\bf The algebra $\CA_1$}.
Recall that the algebras  $\CD (A)$ and $\CD(P)$ are
homogeneous subalgebras of the $\Z$-graded algebra $A_{1,x}$.
So, the intersection
$\CA_1 : = \CD (A) \cap \CD(P) = \CD (A) \cap A_1$
is a   homogeneous subalgebra of the algebras
$\CD(A)$, $A_1$ and $A_{1,x}$.
Clearly,
$ \CA_1 = \{ \d \in \CD (P) \; | \; \d \ast A \subseteq A \}$.

\begin{theorem}\label{BB18Nov18}%\marginpar{BB18Nov18}
Let the algebra $A$ be as in Theorem \ref{AB17Nov18}.
\begin{enumerate}
\item $ \CA_1 = \bigoplus_{i \in \Z} D w_i$ where $w_0 := 1$, $w_i = \d_i$, $w_{-i} = a_i \der^{i}$ for $i \geq 1$, and
\[
a_i = \begin{cases}
             \prod_{j=m-i+1}^{m} (h-j) & \text{if} \, \, \,  i =1, \ldots, m-2,  \\
              \prod_{j=2}^{m} (h-j)  & \text{if} \, \, \,  i \geq m-1. \\
             \end{cases}
\]
\item The algebra $\CA_1$ is a finitely generated algebra and the set
 $\{ w_{-m} , w_{-1}=\d_{-1}, h, \d_1 , \ldots, \d_{m-1}, x^m  \}$
 is an algebra generating set, and $w_{-1}=\d_{-1}$.  *** find defining relations ***
 \item
 			\begin{enumerate}
 					\item  For all $ i \geq m$, $w_{-i} w_{-1} = h w_{-i-1}$, $w_{-1} w_{-i} = (h - 1) w_{-1 -i}$ and $[w_{-i}, w_{-1}] = w_{-m -1}$.
 					\item For $i = 1, \ldots, m-1$, $(w_{-1})^i = w_{-i}$.
 					\item ${(w_{-1})}^{m} = h w_{-m}$.
 					\item For all $i \geq 1$, $[ \d_1 , x^i ] = i x^{i+1}$.
 					\item $w_{-1} \d_{1} = h (h-1)(h-m)$ and $\d_i w_{-1} = (h-1)(h-2)(h-m-1)= \s(h(h-1)(h-m))$.
 					\item For $i = 2, \ldots , m-1$, $w_{-1} \d_{i} = h (h-m) \d_{i-1}$ and $\d_{i} w_{-1} =  (h-i-1) (h-i-m) \d_{i-1}$.
 			\end{enumerate}
\end{enumerate}
\end{theorem}

{\it Proof}.
$1$.
Notice that $\CD(A)_{+} \subseteq A_1$, and so $\CD(A)_{+} \subseteq \CA_1$.
Now statement 1 follows from the fact that the Weyl algebra
$A_1 = D [ x, \der; \s, h] = \bigoplus_{i \geq 1} D \der^{i} \oplus D \oplus \bigoplus_{i \geq 1} D x^{i} $ is a GWA and
from (\ref{xmiph1})--(\ref{xmiph3}).

$3$.
Straightforward.

$2$.
By statement 1, the set
$G = \{ h , w_i \; | \;  i \in \Z \backslash \{0\} \}$
is a generating set for the algebra $\CA_1$.
By the statements 3(a) and 3(b),
the elements $\{ w_i \; | \;  i \leq -m-1 \}$ and
$ \{ w_{-j} \; | \;  j = 2, \ldots , m-1 \}$ are redundant in $G$.
Similarly, by the statement 3(d), the elements
$\{ w_{i} \; | \;  i \geq m+1 \}$ are also  redundant in $G$,
and statement 2 follows. $\Box$\\




 {\bf The generalized Weyl algebras $\mA$ and $\mB$ such that $\mA \subset \CD (A)\subset \mB \subset T^{-1}\mA =T^{-1} \CD (A)= T^{-1} \mB $.} Let $\mA$ be the subalgebra of $\CD (A)$ which is generated by the elements $\d_{-1}$, $h$ and $\d_1$. By Theorem \ref{AB17Nov18}. (1),
 $\d_{-1}= h(h-m)x^{-1}$ and $\d_1= (h-2)x$, and so the algebra
%\marginpar{ADdd}
\begin{equation}\label{ADdd}
\mA = D[\d_1, \d_{-1}; \s , h(h-1)(h-m)], \;\; D=K[h], \;\; \s (h) = h-1,
\end{equation}
 is a GWA such that $\mA\subset \CA_1$ since $\d_{-1}, h , \d_1\in  \CA_1$ (Theorem \ref{BB18Nov18}.(2)). In particular, the algebra  $\mA = \bigoplus_{i>0}D\d_{-1}^i\oplus \bigoplus_{i\geq 0}D\d_1^i$ is a free left/right $D$-module where the set $\{ \d_{\pm 1}^i\, | \, i\geq 0\}$ is a free basis over $D$.


  The multiplicative submonoid $T= \langle h-i\, | \, i\in \Z\rangle$ of $D$ is a (left and right) denominator set of the algebras $\mA$, $
 \CA_1$, $\CD (A)$ and $A_1$ such that
 %\marginpar{BBT}
\begin{equation}\label{BBT}
T^{-1}\mA \simeq T^{-1}\CA_1  \simeq T^{-1} \CD (A) \simeq  T^{-1}A_1=:  \mB = T^{-1}D[x,x^{-1}; \s ], \; T^{-1}D=K[h, (h-i)^{-1}]_{i\in \Z}, \; \s (h) = h-1.
\end{equation}
This follows from the explicit descriptions of the free bases over $D$ of the algebras $\mA$, $\CA_1$, $\CD (A)$ and $A_1=\bigoplus_{i>0}D\der^i\oplus \bigoplus_{i\geq 0}Dx^i$ (Theorem \ref{AB17Nov18}, Theorem \ref{BB18Nov18}). Notice that the algebra $\mB = T^{-1}D[x,x^{-1}; \s , 1] $ is a GWA where the ring $T^{-1}D$ is a Dedekind ring.

 Similarly, the multiplicative set $D\backslash \{ 0\}$ is a (left and right) denominator set of the algebras $\mA$, $
 \CA_1$, $\CD (A)$ and $A_1$ such that
 %\marginpar{BBT1}
\begin{equation}\label{BBT1}
D^{-1}\mA \simeq D^{-1}\CA_1  \simeq D^{-1} \CD (A) \simeq  D^{-1}A_1=:   B = K(h)[x,x^{-1}; \s ], \;\;\s (h) = h-1,
\end{equation}
where $D^{-1}\mA$ denotes the localization  $(D\backslash \{ 0\})^{-1}\mA$ of the algebra $\mA$ at $D\backslash \{ 0\}$, and $K(h)$ is the field of rational functions in the variable $h$ over the field $K$. We have the following diagram of algebras where the vertical lines denote containments of the algebras:

\begin{figure}[h]
\centering
\begin{tikzpicture}
\node (A)  at (0,3) {$B= D^{-1}\mA = D^{-1}\CA_1  = D^{-1} \CD (A) =  D^{-1}A_1 = K(h)[x,x^{-1}; \s ]$};
\node (B)  at (0,2){$\mB= T^{-1}\mA = T^{-1}\CA_1  = T^{-1} \CD (A) =  T^{-1}A_1 = T^{-1}D[x,x^{-1}]$};
\node (C) at (-1,1) {$\CD(A)$};
\node (D)  at (1,1){$A_1$};
\node (E)   at (0, 0){ \qquad \qquad \qquad  $\CA_1=\CD (A) \cap A_1$};
\node (F)   at (-1, 0){$\CA$};
\node (G)   at (0,-1){$\mA$};
\draw [semithick, -] (A)--(B);
\draw [semithick, -] (B)--(C);
\draw [semithick, -] (B)--(D);
\draw [semithick, -] (C)--(F);
\draw [semithick, -] (E)--(G);
\draw [semithick, -] (C)--(E);
\draw [semithick, -] (D)--(E);
\end{tikzpicture}
%\caption{}%\label{AADBB}
\end{figure}

We will see that the properties of the algebra $\CD (A)$ are a mixture of  properties of the GWAs $\mA$ and $\mB$. Theorem \ref{A24Jan20} and Theorem \ref{B24Jan20} are about some properties of the algebras $\mA$ and $\mB$.

\begin{theorem}\label{A24Jan20}%\marginpar{A24Jan20}
\begin{enumerate}
\item The algebra $\mA$ is a finitely generated, central, non-simple Noetherian domain with $\GK (\mA ) = 2$.
\item (\cite[Theorem 1.6]{Bav-1996THM}) $\gldim (\mA) =2$.
\item (\cite[Theorem 2]{Bav-UkrMathJ-92}) All nonzero left ideals of the algebra $\mA$ are co-finite ($\dim_K(\mA / I)<\infty$).
\item (\cite[Theorem 2]{Bav-UkrMathJ-92}) $\Kdim (\mA )=1$.
\item (\cite[Theorem 4]{Bav-UkrMathJ-92}) In $\mA$ there are  only finitely many nonzero ideals.
\item (\cite[Theorem 1]{Bav-UkrMathJ-92}) Up to isomorphism, there only tow simple finite dimensional $\mA$-modules: $L_1=\mA/\mA(\d_{-1}, h-1 ,\d_1)$, $\dim_K(L_1)=1$ and
 $L_{m-1}=\mA/\mA(\d_{-1}^{m-1}, h-m ,\d_1)$, $\dim_K(L_{m-1})=m-1$.
 \item (\cite[Theorem 3.3]{Bav-GWArep}) The category of finite dimensional modules is not semisimple.
 \item (\cite[Theorem 6]{Bav-GWArep}) For all simple $\mA$-modules $M$ and $N$,  the vector spaces $\Ext_{\mA}^i(M,N)$ and   ${\rm Tor}^{\mA}_i(M,N)$ are finite dimensional for all $i$.
 \item (\cite[Theorem 4]{Bav-GWArep}) Let $M$ be a simple $\mA$-module and $q\in \mA \backslash K$, then the kernel and cokernel of the linear map $q_M :M\ra M$, $ m\mapsto qm$ are finite dimensional.

\end{enumerate}
\end{theorem}




\begin{theorem}\label{B24Jan20}%\marginpar{B24Jan20}
\begin{enumerate}
\item The algebra $\mB$ is a finitely generated, central, simple Noetherian domain with $\GK (\mB ) = 2$.
\item (\cite[Theorem 1.6]{Bav-1996THM}) $\gldim (\mB) =1$.
\item (\cite[Theorem 2]{Bav-UkrMathJ-92}) All nonzero left ideals of the algebra $\mB$ are co-finite ($\dim_K(\mB / I)<\infty$).
\item (\cite[Theorem 2]{Bav-UkrMathJ-92}) $\Kdim (\mB )=1$.
\item (\cite[Theorem 1, Theorem 5]{Bav-UkrMathJ-92}) All simple $\mB$-modules are infinite dimensional.
\end{enumerate}
\end{theorem}


{\bf Every proper factor module of $\CD (A)$ has finite length and the Krull dimension of $\CD (A)$.}
 Recall that the algebra $\CD (A)$ is a finitely generated over its subalgebra $\CA$. Proposition shows that the subalgebra $\CA$ of $\CD (A)$ is large in the sense that it meets every nonzero left ideal of the algebra $\CD (A)$.

\begin{proposition}\label{D29Jan20}%\marginpar{D29Jan20}
For all   nonzero left ideals $I$  of the algebra $\CD (A)$, $\CA \cap I\neq 0$.
\end{proposition}

{\it Proof}. The Gelfand-Kirillov dimensions of the domains $\CD (A)$ and $\CA$ is 2.  By Theorem \ref{AB18Nov18}, the algebra $\CD (A)$  is a finitely generated module over its subalgebra $\CA$. Hence, $2=\GK (\CA ) \leq \GK_\CA (\CD (A))\leq \GK (\CA ) =2$, and so $\GK_\CA (\CD (A))=2$. Then, by \cite[Proposition 8.3.5]{MR}, $$\GK_{\CA}(\CD (A)/I)<\GK_{\CA}(\CD (A))-1=2-1=1.$$
Hence, $\CA \cap I\neq 0$  since $\GK (\CA )=2>1=\GK_{\CA}(\CD (A)/I)$.  $\Box $






\begin{theorem}\label{C29Jan20}%\marginpar{C29Jan20}
For all   nonzero left ideals $I$  of the algebra $\CD (A)$,  the $\CD (A)$-module $\CD (A)/I$ has finite length.
\end{theorem}

{\it Proof}.  By \cite[Theorem 2.1]{Bav-GWArep}, for all   nonzero left ideals $I'$  of the algebra $\CA$,  the $\CA$-module $\CA /I'$ has finite length.  By Theorem \ref{AB18Nov18}, the algebra $\CD (A)$ is a finitely generated $\CA$-module. Now, the theorem follows from Proposition \ref{D29Jan20}.  $\Box $



\begin{theorem}\label{E29Jan20}%\marginpar{E29Jan20}
The Krull dimension of the algebra $\CD (A)$ is 1.
\end{theorem}

{\it Proof}. The theorem follows at once from Theorem \ref{C29Jan20}. $\Box$

























%%%%%%%%%%%%%%%%%% SECTION 3 %%%%%%%%%%%%%%%%%%%%%%%%



\section{Classification of simple $\CD (A)$-modules}\label{SIMMOD}%\marginpar{SIMMOD}

The aim of this section is to classify simple $\CD (A )$-modules
 where $A= K+\sum_{i\geq m} Kx^i$ (Theorem \ref{DTor27Jan20} and Theorem \ref{DTFr27Jan20}). They are partitioned in two (disjoint) sets: $D$-torsion and $D$-torsion free.
 The simple $\CD (A)$-modules in each of the two sets are classified (Theorem \ref{DTor27Jan20} and Theorem \ref{DTFr27Jan20}).
% For simplicity, we assume that the ground field $K$ is an algebraically closed field of characteristic zero.



At the beginning of the section we recall a classification of simple modules over a generalized Weyl   $A=D(\s, a)=D[x,y;\s,a]$ where $D$ is  a (commutative) Dedekind domain with some extra condition on the automorphism that is satisfied for our GWAs. In all the papers we cite below these algebras are denoted by `$A$', we hope that this notation will not lead to confusion.

For an algebra $A$, we denote by $\widehat{A}$ the set of isomorphism classes of simple $A$-modules.  For an $A$-module $M$,  we denote by $[M]$ its isomorphism class. If $P$ is an isomorphism invariant  property of simple modules  (e.g., `being weight') then $\widehat{A}\,(P)$ stands for the set of all isomorphism classes of simple $A$-modules that satisfy $P$.\\



{\bf Classification of simple $A$-modules where $A=D(\s, a)$ and $D$ is a Dedekind ring}.
Let $A=D(\s, a)=D[x,y;\s,a]$ be a GWA such that $D$ is a Dedekind ring, $a \neq 0$, and the automorphism $\s$ of $D$ satisfies the condition:\\

 (*)  $\s^i(\gp) \neq \gp$ {\em  for all $i \in\Z \setminus \{0\}$ and all maximal ideals $\gp$ of} $D$.\\

 \textbf{Example}. The Weyl algebra $A_1=K[h][x, \der ; \s ,h]$ is an example of the GWA $A$.\\

\textbf{Example}. $A=K[h] ( \s,a)$ where $\s (h)=h-1$ and $K$ is a field of characteristic zero. In particular, the algebras $\mA$ is  of this type, see (\ref{ADdd}). A classification of simple $K[h](\s,a)$-modules is given in \cite{Bav-UkrMathJ-92,Bav-GWArep}.\\

\textbf{Example}. The GWA $\mB$ is an example of the GWA $A$, see (\ref{BBT}).\\

The set $S:=D\setminus \{ 0\}$ is an Ore set of the domain $A$. So, a simple $A$-module $M$  is either $D$-{\em torsion} ($S^{-1}M=0$) or   $D$-{\em torsion free} ($S^{-1}M \neq 0$). In the second case, the $S^{-1}A$-module $S^{-1}M$ is simple.

Let us recall a classification of simple $A$-modules for the algebra $A=D(\s,a)$, see \cite{Bav-GWArep,Bav-UkrMathJ-92,Bav-GWAKernel-1993} for details. Clearly,
%\marginpar{AADT}
\begin{equation}\label{AADT}
\widehat{A}= \widehat{A}\,(D\text{-}{\rm torsion}) \,\,\coprod\,\,\widehat{A}\,(D\text{-}{\rm torsion \; free}).
\end{equation}





\textbf{The set $\widehat{A}\,(D\text{-}{\rm torsion})=\widehat{A}\,({\rm weight})$}.
The group $\langle \s \rangle \simeq \Z$ acts freely on the set $\Max\,(D)$ of maximal ideals of the Dedekind ring $D$. For each maximal ideal $\gp$ of $D$, $\CO(\gp)=\{ \s^i(\gp)\,|\, i \in \Z \}$ is its orbit. We use the bijection $\Z \rightarrow \CO(\gp)$, $i \mapsto \s^i(\gp)$, to define the order $\leqslant$ on each orbit $\CO(\gp)$: $\s^i(\gp) \leqslant \s^j(\gp)$ iff $i \leqslant j$. A maximal ideal of $D$ is called \emph{marked} if it contains the element $a$. There are only finitely many marked ideals. An orbit $\CO$ is called \emph{degenerated} if it contains a marked ideal. Marked ideals, say $\gp_1 < \cdots <\gp_s$, of a degenerated orbit $\CO$ partition it into $s+1$ parts,
%\marginpar{GGG}
\begin{equation}\label{GGG}
 \G_1=(-\infty, \gp_1], \,\,\,\,  \G_2=(\gp_1, \gp_2], \ldots, \G_s=(\gp_{s-1}, \gp_s], \,\,\,\,\G_{s+1}=(\gp_s, \infty).
\end{equation}
Two ideals $\gp, \gq \in \Max\,(D)$ are called \emph{equivalent} $\gp \sim \gq$ if they belong either to a non-degenerated orbit or to some $\G_i$. We denote by $\Max\,(D)/\!\!\sim $ the set of equivalence classes in $\Max\,(D)$.

An $A$-module $V$ is called \emph{weight} if $V=\bigoplus_{\gp \in \Max\,(D)}V_{\gp}$ where $V_{\gp}=\{v \in V\, |\, \gp v =0 \} =$ the sum of all simple $D$-submodules of $V$ which are isomorphic to $D/\gp$. The set $\Supp\,(V)=\{ \gp \in \Max\,(D)\,|\, V_{\gp} \neq 0 \}$ is called the \emph{support} of $V$, elements of $\Supp\,(V)$ are called \emph{weights} and $V_{\gp}$ is called the \emph{component} of $V$ of weight $\gp$. Clearly, an $A$-module is weight iff it is a semisimple $D$-module. Clearly,
%\marginpar{AD=AW}
\begin{equation}\label{AD=AW}
\widehat{A}\,(D\text{-}{\rm torsion})=\widehat{A}\,({\rm weight}),
\end{equation}
i.e., a simple $A$-module is $D$-torsion iff it is weight.

\begin{theorem}\label{AA27Feb16} %\marginpar{AA27Feb16}
\cite{Bav-UkrMathJ-92,Bav-GWArep,Bav-GWAKernel-1993}
{\sc (Classification of simple $D$-torsion/weight $A$-modules)}  \\
The map $\Max\,(D)/\!\!\!\sim  \,\, \rightarrow \widehat{A}\,(D\text{-}{\rm torsion})$, $\G \mapsto [L(\G)]$, is  a bijection with the inverse $[M] \mapsto \Supp\,(M)$ where
\begin{enumerate}
\item If $\G$ is  a non-degenerated orbit then $L(\G)=A/A\gp$ where $\gp \in \G$.
\item If $\G=(-\infty, \gp]$ then $L(\G)=A/A(\gp, x)$.
\item If $\G=(\s^{-n}(\gp), \gp]$ for some $n \geqslant 1$ then $L(\G)=A/A(y^n, \gp ,x)$. The $D$-length of $L(\G)$ is $n$.
\item If $\G=(\gp, \infty)$ then $L(\G)=A/A(\s(\gp), y)$.
\end{enumerate}
\end{theorem}



\textbf{The set $\widehat{A}\,(D\text{-}{\rm torsionfree})$}.
For elements $\alpha, \beta \in D$, we write $\alpha < \beta$ if $\gp < \gq$ for all $\gp, \gq \in \Max\,(D)$ such that $\CO(\gp)=\CO(\gq)$, $\alpha \in \gp$ and $\beta \in \gq$. (We write also $\alpha < \beta$ if there are no such ideals $\gp$ and $\gq$). Recall that the GWA $A=\bigoplus_{i \in \Z}A_i$ is a $\Z$-graded algebra where $A_i=Dv_i=v_i D$, $v_0=1$, $v_i=x^i$ and $v_{-i}=y^i$ for all $i\geq 1$.\\

\emph{Definition}, \cite{Bav-UkrMathJ-92,Bav-GWArep,Bav-GWAKernel-1993}.
An element $b=v_{-m}\beta_{-m}+v_{-m+1}\beta_{-m+1}+\cdots +\beta_0 \in A$ (where $m \geqslant 1$, all $\beta_i \in D$ and $\beta_{-m}, \beta_0 \neq 0$) is called a \emph{normal} element if $\beta_0 < \beta_{-m}$ and $\beta_0 < a$.\\


The set $S:=D\setminus \{ 0\}$ is an Ore set of the domain $A$. Let $k:=S^{-1}D$ be the field of fractions of $D$. The algebra $B:=S^{-1}A=k[x, x^{-1};\s]$ is a skew Laurent polynomial ring which is a (left and right) principle ideal domain. So, any simple $B$-module is of type $B/Bb$ for some irreducible element $b$ of $B$. Two simple $B$-modules are isomorphic, $B/Bb \simeq B/Bc$, iff the elements $b$ and $c$ are \emph{similar} (i.e., there exists an element $d \in B$ such that $1$ is the greatest common right divisor of $c$ and $d$, and $bd$ is a least common left multiple of $c$ and $d$).


\begin{theorem}\label{BA27Feb16} %\marginpar{BA27Feb16}
\cite{Bav-UkrMathJ-92,Bav-GWArep,Bav-GWAKernel-1993} {\sc (Classification of simple $D$-torsionfree $A$-modules)}\\
$\widehat{A}\,(D\text{-}{\rm torsionfree})=\{ [M_b:=A/A\cap Bb]\,|\,b$ is a normal irreducible element of $B\}$. The $A$-modules $M_b$ and $M_{b'}$ are isomorphic iff the elements $b$ and $b'$ are similar.
\end{theorem}

For all nonzero elements $\alpha, \beta \in D$, the $B$-modules $S^{-1}M_b$ and $S^{-1}M_{\beta b \alpha^{-1}}$ are isomorphic. If an element $b=v_{-m}\beta_{-m}+\cdots +\beta_0$ is irreducible in $B$ but not necessarily normal the next lemma shows  that there are explicit elements $\alpha$ and $\beta$ such that the element $\beta b \alpha^{-1}$ is normal and irreducible in $B$.

\begin{lemma}\label{a27Feb16} %\marginpar{a27Feb16}
\cite[Lemma 13]{Bav-UkrMathJ-92} {\sc (Normalization procedure)} \\
Given an element $b= v_{-m} \beta_{-m}+\cdots+\beta_0 \in A$ where $m \geqslant 1$, all $\beta_i \in D$ and $\beta_{-m}, \beta_0 \neq 0$. Fix a natural number $s \in \N$ such that $\s^{-s} (\beta_0) < \beta_{-m}$, $\s^{-s}(\beta_0)< \beta_0$ and $\s^{-s}(\beta_0)<a$. Let $\alpha=\prod_{i=0}^s \s^{-i}(\beta_0)$ and $\beta=\prod_{i=1}^{s+m}\s^{-i}(\beta_0)$. Then the element $\beta b \alpha^{-1}$ is a normal element which is called a normalization of $b$ and denoted $b^{\rm norm}$ (we can always assume that $s$ is the least possible).
\end{lemma}

{\bf The algebra $A=K+\sum_{i\geq m} Kx^i$ is a simple weight $\CD (A)$-module.} Clearly, $A=\sum_{i\in E}Kx^i$  where $E:=\{ 0,  m, m+1, \ldots \}$.   Recall that  by the very definition of the algebra $\CD (A)$ of differential operators on $A$, the algebra $A$ is a left $\CD (A)$-module and the action of  an element $\d \in \CD (A)$ on an element $a\in A$ is denoted either by $\d *(a)$ or $\d (a)$. For all $i\in E$, $h*x^i=(i+1)x^i$. This implies that the $\CD (A)$-module $A$ is a weight $\CD (A)$-module with $\Supp (A)=\{ (h-i-1)\, | \, i\in E\}$. In particular the $\CD (A)$-module $A$ is $D$-torsion.
If follows from the equalities
%\marginpar{Eldd}
\begin{equation}\label{Eldd}
\d_1= (h-2)x, \; \d_{-1}= h(h-m)x^{-1}, \; \d_{-1}\d_1= h(h-1)(h-m)\; {\rm and}\; \d_1\d_{-1}=  (h-1)(h-2)(h-m-1)
\end{equation}
that the maps
\begin{eqnarray*}
\d_1 &:& Kx^i\ra Kx^{i+1}, \;\;\;  p\mapsto \d_1*p, \; \;\;\; i\geq m, \\
 \d_{-1} &:& Kx^{i+1}\ra Kx^i, \;\;\; p\mapsto \d_{-1}*p, \; \; i\geq m,
\end{eqnarray*}
are bijections. Similarly, it follows from the equalities
$\d_{-m}\d_m =(h+m-1) (h-2) \cdots (h-m)$ and $\d_m \d_{-m}=(h-1) (h-m-2) \cdots (h-2m)$ that the maps
\begin{eqnarray*}
\d_m &:& K\ra Kx^m, \;\;  p\mapsto \d_m*p, \\
 \d_{-m} &:& Kx^m\ra K, \; \; p\mapsto \d_{-m}*p,
\end{eqnarray*}
are bijections. Therefore, the algebra $A$ is a simple weight $\CD (A)$-module  with $\Supp (A)=\{ (h-i-1)\, | \, i\in E\}$. \\

{\bf Classification of simple weight $\CD (A)$-modules with support that belongs to the orbit $\OO (h)$.} The ideal $(h) = Dh$ is a maximal ideal of the polynomial algebra $D=K[h]$ with $D/(h) =K$. Let $\OO (h)=\OO ((h))=\{ \s^i (h)=(h-i) \, | \, i\in \Z\}$ be its $\s$-orbit. We will see that (up to isomorphism) there are only two simple weight $\CD (A)$-modules with support in $\OO (h)$: the algebra $A$ and a `complementary' module $A'$ which we are going to define. Furthermore, $\supp (A') = \OO (h) \backslash \Supp (A)$.

The polynomial algebra $K [x]$ has the canonical structure of the left $A_1$-module.
Namely, $K [x] \simeq A_1 / A_1 \der $; $ x \ast p = x p $ and
$ \der \ast p = \dfrac{d p}{d x}$ for all $p \in P$.
The {\em Laurent polynomial algebra} $L = K [x , x^{-1}]=\bigoplus_{i\in \Z} Kx^i$,
which is the localization of the polynomial algebra $K[x]$ at
$S_x = \{ x^{i} \; | \;  i \geq 0 \}$,
is a left $A_{1,x}$-module.
%Since $\CD (A) \subseteq A_{1,x}$, the algebra $L$ is a left $\CD(A)$-module.
By (\ref{ASAD1}), the Laurent polynomial algebra $L$ is a left module over the algebras $A_{1,x}\simeq S_{x^m}^{-1}\CD (A)$ and $\CD (A)$. One can easily verifies using Theorem \ref{AB17Nov18}, that the subalgebra $A$ is a $\CD (A)$-submodule of $L$. Consider the $\CD (A)$-module,
%\marginpar{Apdef}
\begin{equation}\label{Apdef}
A':=L/A=\bigoplus_{i\in E'}Kx^i, \;\; E':=\Z\backslash E=\{ \ldots , -2,-1, 1, 2, \ldots , m-1\}.
\end{equation}
By (\ref{Eldd}),  the maps
\begin{eqnarray*}
\d_1 &:& Kx^i\ra Kx^{i+1}, \;\;  p\mapsto \d_1*p, \; \;\;\;\; i\in E'\backslash \{ -1, m-1\}, \\
\d_{-1} &:& Kx^{i+1}\ra Kx^i, \;\; p\mapsto \d_{-1}*p, \;\;\;  i\in E'\backslash \{ -1, m-1\},
\end{eqnarray*}
are bijections.
Since $\d_2= (h-3)x^2$ and $\d_{-2}= (h+1) (h-m+1) (h-m) x^{-2}$, we have that
%\marginpar{Eldd1}
\begin{equation}\label{Eldd1}
\d_{-2}\d_2= (h+1) (h-m+1) (h-m) (h-1)
\; {\rm and}\; \d_2\d_{-2}= (h-3) (h-1)(h-m-1)(h-m-2),
\end{equation}
and so   the maps
\begin{eqnarray*}
\d_2 &:& Kx^{-1}\ra Kx, \;\;  p\mapsto \d_2*p, \\
 \d_{-2} &:& Kx\ra Kx^{-1}, \; \; p\mapsto \d_{-2}*p,
\end{eqnarray*}
are bijections.  Therefore, the $\CD (A)$-module   $A'$ is a simple weight $\CD (A)$-module  with $\Supp (A')=\OO (h) \backslash \Supp (A)= \{ (h-i-1)\, | \, i\in E'\}$.

\begin{lemma}\label{a26Jan20}%\marginpar{a26Jan20}
The $\CD (A)$-modules $A$ and $A'$ are the only two (up to isomorphism) simple weight $\CD (A)$-modules with support in the orbit $\OO (h)$.
\end{lemma}

{\it Proof}. Recall that the $\CD (A)$-modules $A$ and $A'$ are non-isomorphic   simple weight $\CD (A)$-modules with support in the orbit $\OO (h)$. Now, the lemma follows at once from the fact that every simple weight module is uniquely determined by its support and that $\OO (h)=  \Supp (A) \coprod \Supp (A')$. $\Box $

Let us collect properties of the $\CD (A)$-modules $A$ and $A'$ in the next two lemmas.

\begin{lemma}\label{b26Jan20}%\marginpar{b26Jan20}
\begin{enumerate}
\item The algebra $A=\bigoplus_{i\in E}Kx^i$ is a simple weight $S_{x^m}$-torsion $\CD (A)$-module with  $\Supp (A)=\{ (h-i-1)\, | \, i\in E\}$  where  $E=\{ 0,  m, m+1, \ldots \}$, and $\End_{\CD (A)}(A)=K$.
\item ${}_{\CD (A)}A\simeq \CD (A) / \CD (A) (h-1, \d_{-1})=\bigoplus_{i\in E} K\d_i\overline{1} $ where $\overline{1}:= 1+\CD (A) (h-1, \d_{-1})$.
\end{enumerate}
\end{lemma}

{\it Proof}. 1. The weight spaces of the weight $\CD (A)$-module $A$  are 1-dimensional,  hence  $\End_{\CD (A)}(A)=K$. The rest of statement 1 have been proven above.

2. The $\CD (A)$-module $W=\CD (A) / \CD (A) (h-1)\simeq \bigoplus_{i\in \Z }K\d_i1^*$ is weight  with $\Supp (W)=\Z$ where $1^*=1+\CD (A) (h-1)$. The map $W\ra A$, $1^*\mapsto \overline{1}$ is a $\CD (A)$-module epimorphism. Hence, ${}_{\CD (A)}A\simeq \CD (A) / \CD (A) (h-1, \d_{-1})$, by Lemma \ref{a26Jan20}.   $\Box $

\begin{lemma}\label{c26Jan20}%\marginpar{c26Jan20}
\begin{enumerate}
\item The algebra $A'=\bigoplus_{i\in E'}Kx^i$ is a simple weight $S_{x^m}$-torsion $\CD (A)$-module with  $\Supp (A')=\OO (h)\backslash \Supp (A)= \{ (h-i-1)\, | \, i\in E'\}$
 where  $E'= \Z \backslash E =  \{ \ldots , -2,-1, 1, 2, \ldots , m-1\}$, and $\End_{\CD (A)}(A')=K$.
\item ${}_{\CD (A)}A'\simeq \CD (A) / \CD (A) (h, \d_1)=\bigoplus_{i\in E'} K\d_i\overline{1}' $ where $\overline{1}':= 1+\CD (A) (h, \d_1)$.
\end{enumerate}
\end{lemma}

{\it Proof}. 1. The weight spaces of the weight $\CD (A)$-module $A'$  are 1-dimensional,  hence  $\End_{\CD (A)}(A')=K$. The rest of statement 1 have been proven above.

2. The $\CD (A)$-module $W'=\CD (A) / \CD (A) h\simeq \bigoplus_{i\in \Z }K\d_i1^o$ is weight  with $\Supp (W)=\Z$ where $1^o=1+\CD (A) h$. The map $W'\ra A'$, $1^o\mapsto \overline{1}'$ is a $\CD (A)$-module epimorphism. Hence, ${}_{\CD (A)}A'\simeq \CD (A) / \CD (A) (h, \d_1)$, by Lemma \ref{a26Jan20}.   $\Box $\\

{\bf Classification of simple $D$-torsion $\CD (A)$-modules.}   Recall that $\mA\subset \CD (A) \subset \mB = T^{-1}\mA=T^{-1}\CD (A)$.
 So, every $\mB$-module is automatically is an $\mA$-module and $\CD (A)$-module. The group $\langle \s \rangle $ acts on the set $\Max (D)$ of maximal ideal of the algebra $D=K[h]$. The field $K$ has characteristic zero and $\s (h) = h-1$. So, every orbit $\OO (\gp ) = \{ \s^i (\gp )\, | \, i\in \Z\}$ contains infinite number of elements where $\gp \in \Max (D)$.
 We denote by $ \Max (D)/\langle \s \rangle $ is the set of all $\s$-orbits in  $ \Max (D)$.


 The algebra $\mB = T^{-1}D[x,x^{-1}; \s , 1]$ is a GWA where $ T^{-1}D$ is a Dedekind ring and the automorphism $\s$ satisfies the condition (*) above.
 Notice that $\Max (T^{-1} D) = \{ T^{-1}\gp \, | \, \gp \in \Max (D)\backslash \OO (h)\} $ where $\OO (h)$ is the $\s$-orbit of the maximal ideal $(h)$ of the algebra $D$, and the map
 $ \Max (D)\backslash \OO (h)\ra \Max (T^{-1} D)$, $\gp\mapsto T^{-1}\gp$
 is a bijection.

 For each orbit $\OO\in \Max (D)/\langle \s \rangle \backslash \{ \OO (h)\}$, we fix its element, say $\gp_\OO$. So, $\OO (\gp_\OO )=\OO $.
 \begin{proposition}\label{A29Jan20}%\marginpar{A29Jan20}

\begin{enumerate}
\item $\widehat{\mB} (T^{-1}D-{\rm torsion})= \{ \mB / \mB \gp_\OO  \, | \, \gp_\OO \in \Max (D)/\langle \s \rangle \backslash \{ \OO (h)\} \}$.
\item The restriction map  $\widehat{\mB} (T^{-1}D-{\rm torsion})\ra  \widehat{\CD (A)} (D-{\rm torsion})$, $M\ra {}_{\CD (A)}M$ is an injection.
\end{enumerate}
\end{proposition}

{\it Proof}. 1. Statement 1 follows at once from Theorem \ref{AA27Feb16} and the fact that the defining element of the GWA $\mB$ is $1$, and so every orbit of the automorphism $\s$ in $\Max (T^{-1}D)$ is not degenerated.

2. Given $[M]\in \widehat{\mB} (T^{-1}D-{\rm torsion})$. By statement 1,  $$M= \mB / \mB \gp =\bigoplus_{i\in \Z} x^{-i}T^{-1}D/T^{-1}D\gp \simeq \bigoplus_{i\in \Z} x^{-i} D/D\gp $$ is a direct sum of non-isomorphic simple $D$-modules  for some $\gp = \gp_\OO \in \Max (D)\backslash \OO (h)$. By  Theorem \ref{AA27Feb16} in case of the GWA $\mA$, the weight $\mA$-module $M$ is simple, hence the $\CD (A)$-module $M$ is simple since $\mA\subset \CD (A)$. $\Box $

In view of Proposition \ref{A29Jan20}.(2), we can write $\widehat{\mB} (T^{-1}D-{\rm torsion}) \subseteq  \widehat{\CD (A)} (D-{\rm torsion})$.

\begin{theorem}\label{DTor27Jan20}%\marginpar{DTor27Jan20}
{\sc (Classification of simple $D$-torsion $\CD (A)$-modules)}
\begin{enumerate}
\item $\widehat{\CD (A)} (D-{\rm torsion})=\{ A,  A'\} \coprod \widehat{\mB} (T^{-1}D-{\rm torsion})$.
\item  $\widehat{\CD (A)} (D-{\rm torsion})=\{ A,  A'\}  \coprod \{ \mB / \mB \gp_\OO  \, | \, \gp_\OO  \in \Max (D)\backslash \OO (h))\}$, $\Supp (\mB / \mB \gp_\OO )= \OO $.
\item For all $[M]\in \widehat{\CD (A)} (D-{\rm torsion})$, $\Supp (M)=\infty$ and $\dim_K(M) = \infty$.
\end{enumerate}
\end{theorem}

{\it Proof}. 1.  Notice that $\Max (D)= \OO (h)\coprod \Max (T^{-1}D) $ where the inclusion $\Max (T^{-1}D) \subset \Max (D)$ is due to the injection $\Max (T^{-1}D)\ra \Max (D)$, $\gm\mapsto D\cap \gm$. Recall that every simple $D$-torsion $\CD (A)$-module is a simple weight $\CD (A)$-module, and vice versa, see (\ref{AD=AW}). Now, statement 1 follows from
Lemma \ref{a26Jan20} and Proposition \ref{A29Jan20}.

2. Statement 2 follows from statement 1 and Proposition \ref{A29Jan20}.

3. Statement 3 follows from statement 2. $\Box $

In order to describe the set of simple $D$-torsion free $\CD (A)$-modules we need to know a classification of  simple weight $\mA$-modules (Theorem \ref{B29Jan20})  and how  simple weight $\CD (A)$-modules with support from the orbit $\OO (h)$ decompose under restriction to the subalgebra $\mA$ of $\CD (A)$ (Lemma \ref{a29Jan20}). 

\textbf{The set $\widehat{\mA}\,(D\text{-}{\rm torsion})=\widehat{\mA}\,({\rm weight})$}. Recall that the algebra $\mA$ is a generalized Weyl algebra $\mA = D[ \d_1, \d_{-1}; \s , a= h(h-1)(h-m)]$  where $D=K[h]$ and $\s (h) = h-1$. The orbit $\OO (h)$ is the only degenerated orbit and the maximal  ideals $(h)<(h-1)<(h-m)$ are the only marked maximal ideals. They partition the orbit $\OO (h)$ into subsets (see (\ref{GGG})):
$$\G_-=(-\infty, (h)], \,\,\,\,  \G_1=((h), (h-1)],\; \; \G_{m-1}=((h-1), (h-m)], \,\,\,\,\G_+=((h-m), \infty). $$

\begin{theorem}\label{B29Jan20} %\marginpar{B29Jan20}
{\sc (Classification of simple $D$-torsion/weight $\mA$-modules)}  \\
The map $\Max\,(D)/\!\!\!\sim  \,\, \rightarrow \widehat{\mA}\,(D\text{-}{\rm torsion})$, $\G \mapsto [L(\G)]$, is  a bijection with the inverse $[M] \mapsto \Supp\,(M)$ where
\begin{enumerate}
\item If $\G\in \Max (D)/\langle \s \rangle\backslash \{ \OO (h)\}$ is  a non-degenerated orbit then $L(\G)=\mA/\mA\gp$ where $\gp \in \G$.
\item If $\G=\G_-=(-\infty, (h)]$ then $L_-:=L(\G_-)=\mA/\mA( h, \d_1)$.
\item If $\G=\G_1, \G_{m-1}$ then $L_1:=L(\G_1)=\mA/\mA(\d_{-1}, h-1 ,\d_1)$ and $L_{m-1}:=L(\G_{m-1})=\mA/\mA(\d_{-1}^{m-1}, h-m ,\d_1)$. These two modules are the only finite dimensional simple $\mA$-modules;  $\dim_K\, L(\G_1)=1$ and   $\dim_K\, L(\G_{m-1})=m-1$.
\item If $\G=\G_+$ then $L_+:=L(\G_+)=\mA/\mA(h-m-1, \d_{-1})$.
\end{enumerate}
\end{theorem}

{\it Proof.} This is a particular case of Theorem \ref{AA27Feb16}. $\Box$\\

 Recall that $\mA \subset \CD (A) \subset \mB$. So every $\mB$-module is also an $\mA$-module and a $\CD (A)$-module (by restriction). Corollary \ref{b29Jan20} shows that the algebras  $\mA$, $ \CD (A)$ and $ \mB$ have the same simple $D$-torsion modules provided their supports do not belong to the orbit $\OO (h)$. For the algebras $R=\mA$, $\CD (A)$, $ \mB$, we denote by $\widehat{R } (D-{\rm torsion}\, | \, \OO )$ the set of simple $D$-torsion $R$-modules with support disjoint from $\OO (h)$.


  \begin{corollary}\label{b29Jan20}%\marginpar{b29Jan20}
$\widehat{\mA } (D-{\rm torsion}\, | \, \OO )=\widehat{\CD (A)} (D-{\rm torsion}\, | \, \OO )=\widehat{\mB } (D-{\rm torsion}\, | \, \OO )=\{ \mB / \mB \gp_\OO  \, | \, \gp_\OO  \in \Max (D)\backslash \OO (h)\}$ and  $\Supp (\mB / \mB \gp_\OO )= \OO $.
\end{corollary}

{\it Proof}.  The corollary follows from the classifications of simple $D$-torsion modules for the algebras $ \mA$, $\CD (A)$ and $ \mB$ (Theorem \ref{DTor27Jan20} and  Theorem \ref{B29Jan20}). $\Box$

By Lemma   \ref{a26Jan20}, the $\CD (A)$-modules $A$ and $A'$ are the only two (up to isomorphism) simple weight $\CD (A)$-modules with support in the orbit $\OO (h)$. Lemma \ref{a29Jan20} shows that these modules are semisimple $\mA$-modules of length 2.

\begin{lemma}\label{a29Jan20}%\marginpar{a29Jan20}
\begin{enumerate}
\item ${}_{\mA} A= L_1\oplus L_+$ is a direct sum of simple weight $\mA$-modules  where the $\mA$-modules  $L_1$ and $ L_+$ are defined in Theorem \ref{B29Jan20}.(3,4).
\item ${}_{\mA} A'=  L_-\oplus L_{m-1}$   is a direct sum of simple weight $\mA$-modules where the $\mA$-modules  $L_1$ and $ L_+$ are defined in Theorem \ref{B29Jan20}.(2,3).
\end{enumerate}
\end{lemma}

{\it Proof}. 1. Recall that ${}_{\CD (A)} A=K+\sum_{i\geq m} Kx^i$. Then ${}_\mA K\simeq L_1$ and ${}_\mA \Big( \sum_{i\geq m} Kx^i\Big)\simeq L_+$.  Hence, ${}_{\mA} A= L_1\oplus L_+$ since $\d_1*K=0$, $\d_{-1}*K=0$ and $\d_{-1}*x^m=0$.

2. Similarly,  ${}_{\CD (A)} A'=\Big(\sum_{i\leq -1} Kx^i\Big)\oplus \Big(\sum_{1\leq i\leq m-1} Kx^i\Big)$. Then $_\mA \Big(\sum_{i\leq -1} Kx^i\Big)\simeq L_-$ and ${}_\mA \Big(\sum_{1\leq i\leq m-1} Kx^i\Big)\simeq L_{m-1}$. Hence,  ${}_{\mA} A'= L_{m-1}\oplus L_-$  since $\d_1*x^{-1}=0$, $\d_{-1}*x=0$ and $\d_1*x^{m-1}=0$.
$\Box $ 

{\bf Classification of simple $D$-torsion free  $\CD (A)$-modules.}
Recall that the algebra $\mA$ is a GWA $\mA = D[\d_1, \d_{-1}; \s , a=h(h-1)(h-m)]$. In order to stress that we consider `normal'  elements for the GWA $\mA$ we say `$\mA$-{\em normal}', see Theorem \ref{DTFr27Jan20}, i.e.   an element $b=\d_{-1}^m\beta_{-m}+\d_{-1}^{m-1}\beta_{-m+1}+\cdots + \beta_0 \in \mA$ (where $m \geqslant 1$, all $\beta_i \in D$ and $\beta_{-m}, \beta_0 \neq 0$) is called an $\mA$-{\em normal} element if $\beta_0 < \beta_{-m}$ and $\beta_0 < a$.

\begin{theorem}\label{DTFr27Jan20}%\marginpar{DTFr27Jan20}
{\sc (Classification of simple $D$-torsion free $\CD (A)$-modules)}
$\widehat{\CD (A)}\,(D\text{-}{\rm torsion\; free})=\{ [M_b:=\CD (A)/\CD ( A)\cap Bb]\,|\,b$ is an $\mA$-normal irreducible element of $B\}$. The $\CD (A)$-modules $M_b$ and $M_{b'}$ are isomorphic iff the elements $b$ and $b'$ are similar.
\end{theorem}

{\it Proof}. Let $\CR$ be the RHS of the equality in the theorem.

(i) $\CR \subseteq \widehat{\CD (A)}\,(D\text{-}{\rm torsion\; free})$:
Given $M_b:=[\CD (A)/\CD ( A)\cap Bb]\in \CR$ where $b$ is an $\mA$-normal irreducible element of $B$. We have to prove that $M_b\in \widehat{\CD (A)}\,(D\text{-}{\rm torsion\; free})$. By the very definition the $\CD (A)$-module $M_b$ is $D$-torsion free (since $M_b\subseteq B/ Bb$). By Theorem \ref{C29Jan20}, the $\CD (A)$-module $M_b$ has finite length. It remains to show that the $\CD (A)$-module $M_b$ is simple. Suppose that this is not the case, i.e. the left ideal $\CD (A) \cap Bb$ of the algebra $\CD (A)$  is not a maximal left ideal, we seek a contradiction. Then there is an element $\alpha \in D\backslash K$ such that the left ideal  $\CD (A) \cap Bb$ is properly contained in the left ideal $D\alpha +\CD (A) \cap Bb \neq \CD (A)$. Hence, let $W$ be a simple weight $\CD (A)$-factor module of the $\CD (A)$-module $\CD (A)/ (D\alpha +\CD (A) \cap Bb )$. In particular the action of the element $b\in \mA \subseteq \CD (A)$ has nonzero kernel. By Corollary \ref{b29Jan20} and   Lemma \ref{a29Jan20}, the weight  $\mA$-module $W$ is either simple or a direct sum of two  simple weight $\mA$-modules.
 Hence, the action of the element $b$ is not injective on a simple $\mA$-submodule of $W$, this contradicts to  \cite[Lemma 3.7]{Bav-GWArep} since the element $b$ is $\mA$-normal, a contradiction.

(ii) $\CR \supseteq \widehat{\CD (A)}\,(D\text{-}{\rm torsion\; free})$:
Let $M\in \widehat{\CD (A)}\,(D\text{-}{\rm torsion\; free})$. We have to show that $M\simeq M_b$ for some $\mA$-normal  irreducible element $b$  of $B$. The $B$-module $D^{-1}M$ is simple. Hence, $M\simeq M_b$ for some irreducible element of the algebra $B$. Since $D^{-1}\mA = B$ we may assume that $b=\d_{-1}^m\beta_{-m}+\d_{-1}^{m-1}\beta_{-m+1}+\cdots + \beta_0$ with all $\beta_i\in D$,  $\beta_{-m}\neq 0$ and $\beta_0\neq 0$.   By Lemma \ref{a27Feb16},   we may assume that the element $b$ is $\mA$-normal since the $B$-modules $B/Bb$ and $ B/B\beta b\alpha =  B/Bb \alpha$ are isomorphic (via the map $u\mapsto u\alpha $), and the statement (ii) follows. $\Box $
  %%%%%%%%%%%%%%%%%% SECTION 5 %%%%%%%%%%%%%%%%%%%%%%%%

\section{The algebras  $\CDAm$ where $m\in \N^n$}\label{CDAMN}%\marginpar{CDAMN}

In this section, properties of the algebras $\CDAm$ of differential operators  are studied where $m\in \N^n$. Proofs of Theorems \ref{GenProp30Jan20}--\ref{Kdim30JAn20} are given. The key idea of the proofs is to use properties of the generalized Weyl algebras $\CA (m)$ of rank $n$.

{\bf Generalized Weyl algebras of rank $n$, \cite{Bav-GWA-FA-91}-\cite{Bav-Bav-Ideals-II-93}.} Let $D$ be a ring, $\sigma=(\sigma_1,\ldots  ,\sigma_n)$   an $n$-tuple of
commuting automorphisms of $D$,  $a=(a_1,\ldots  ,a_n)$   an $n$-tuple  of elements of  the centre
$Z(D)$  of $D$ such that $\sigma_i(a_j)=a_j$ for all $i\neq j$. The {\bf  generalized Weyl algebra} $A=D(\sigma,a)=D[x, y; \sigma,a]$ of rank  $n$  is  a  ring  generated
by $D$  and    $2n$ indeterminates $x_1, \ldots ,x_n, y_1,\ldots  ,y_n$
subject to the defining relations:
$$y_ix_i=a_i,\;\; x_iy_i=\sigma_i(a_i),\;\; x_id=\sigma_i(d)x_i,\;\;{\rm and}\;\;  y_id=\sigma_i^{-1}(d)y_i\;\; {\rm for \; all}\;\; d \in D,$$
$$[x_i,x_j]=[x_i,y_j]=[y_i,y_j]=0, \;\; {\rm for \; all}\;\; i\neq j,$$
where $[x, y]=xy-yx$. We say that  $a$  and $\sigma $ are the  sets  of
{\it defining } elements and automorphisms of the GWA $A$, respectively.

The GWA $A=\bigoplus_{\alpha\in \Z^n}A_\alpha$ is a $\Z^n$-graded algebra ($A_\alpha A_\beta \subseteq A_{\alpha +\beta}$ for all elements $\alpha , \beta \in \Z^n$) where $A_\alpha = Dv_\alpha = v_\alpha D$, $v_\alpha =v_{\alpha_1}(1) \t\cdots \t v_{\alpha_n}(n)$, $v_{m}(i):=x_i^m$ and  $v_{-m}(i):=y_i^m$ for  all $m\geq 1$, and $v_0(i):=1$. 

{\bf Example.} Let $D_i[x_i, y_i; \s_i , a_i]$ be GWAs of rank 1 over a field $K$ where $i=1, \ldots , n$. Then their tensor product over the field $K$,
$$\bigotimes_{i=1}^nD_i[x_i, y_i; \s_i , a_i]=D[x,y; \s , a],$$
is a GWA of rank $n$ where the $D=\bigotimes_{i=1}^n D_i$, $\s = (\s_1, \ldots , \s_n)$ and $a= (a_1, \ldots , a_n)$. The $\Z^n$-grading of the GWA $D[x,y; \s , a]$ of rank $n$ is the tensor product of $\Z$-gradings of the tensor components/GWAs of rank 1.

{\bf Example.}  The $n$'th Weyl algebra
$A_n=A_n(K)$ is a generalized Weyl algebra
 $A=D_n[x, y; \s ;a]$ of rank $n$ where
$D_n=K[h_1,...,h_n]$ is a polynomial algebra   in $n$ variables with coefficients in $K$, $\s = (\s_1, \ldots , \s_n)$ where $\s_i(h_j)=h_j-\delta_{ij}$,   $\d_{ij}$ is the Kronecker delta function and $a=(h_1, \ldots , h_n)$.  The map
$$A_n\ra A,\;\; x_i\mapsto  x_i,\;\; \der_i \mapsto y_i   ,\;\;  i=1,\ldots ,n,$$ is an algebra  isomorphism (notice that $\der_ix_i\mapsto h_i$). In particular, the GWA $A_n=\bigoplus_{\alpha \in \Z^n}D_nv_\alpha$ is a $\Z^n$-graded algebra where $v_\alpha =v_{\alpha_1}(1) \t\cdots \t v_{\alpha_n}(n)$, $v_{m}(i):=x_i^m$ and  $v_{-m}(i):=\der_i^m$ for all $m\geq 1$, and $v_0(i):=1$. 

{\bf Generators and defining relations for the algebra $\CDAm$.} {\bf Proof of Theorem \ref{GenRel30Jan20}.}

1. The set $S_{n,x} :=\{ \prod_{i=1}^nx_i^{n_i} \; | \;  n_i \geq 0 \}$
(resp., $S_{n,x^m} := \{\prod_{i=1}^nx_i^{m_in_i} \; | \;   n_i \geq 0 \}$)
is a multiplicative set of the polynomial algebra $P_n=K[x_1, \ldots , x_n] $
(resp., $P_n$ and $\Am$).
Clearly,
%\marginpar{SxmPAn1}
\begin{equation}\label{SxmPAn1}
K[x,x^{-1}]:=K[x_1^{\pm 1}, \ldots , x_n^{\pm n}] = S_{n,x}^{-1} P_n = S_{n,x^m}^{-1} P_n = S_{n,x^m}^{-1} \Am .
\end{equation}
The set $S_{n,x}$
(resp., $S_{n,x^m}$)
is an Ore set of the Weyl algebra $A_n$
(resp., of $A_n$ and $\CDAm$)
and
%\marginpar{ASADn1}
\begin{equation}\label{ASADn1}
A_{n,x} := S_{n,x}^{-1} A_n = S_{n,x^m}^{-1} A_n = S_{n,x^m}^{-1} \CD(P_n) \simeq  \CD (S_{n,x^m}^{-1} P_n) \stackrel{(\ref{SxmPAn1})} = \CD (S_{n,x^m}^{-1} \Am ) \simeq  S_{n,x^m}^{-1} \CDAm .
\end{equation}
Recall that the Weyl algebra
$A_n = D_n [x, \der ; \s , a  ]$ is a GWA of rank $n$, see above.
In particular, the Weyl algebra
$A_n = \bigoplus_{\alpha \in \Z^n} D_n v_\alpha$ is a $\Z^n$-graded algebra.

Since the elements of the Ore set $S_{n,x}$
%and $S_{x^m}$
are homogeneous elements of the algebra $A_n$,
the localized algebra $A_{n,x}$ is also a $\Z^n$-graded algebra
%\marginpar{Anx}
\begin{equation}\label{Anx}
A_{n,x} = \bigoplus_{\alpha \in \Z^n} D_n x^\alpha =D_n [x_1^{\pm 1},  \ldots ,  x_n^{\pm 1}; \s_1, \ldots , \s_n]
\end{equation}
which is a skew Laurent polynomial algebra where $D_n=K[h_1, \ldots , h_n]$, $h_i=\der_i x_i$ and  $x^\alpha = \prod_{i=1}^n x_i^{\alpha_i}$.
By (\ref{ASADn1}),
$\CDAm = \{ \d \in A_{n,x} \; | \; \d \ast \Am \subseteq \Am  \}$.

Since the algebra $\Am$ is a $\Z^n$-graded subalgebra of the polynomial algebra $P_n$, the algebra $\CDAm$ is also $\Z^n$-graded,
%\marginpar{ASAD1n1}
\begin{equation}\label{ASAD1n1}
\CDAm = \bigoplus_{\alpha \in \Z^n} \CDAm_{[\alpha]}  \;\; {\rm  where} \;\;  \CDAm_{[\alpha]} = \CDAm \cap D_n x^\alpha  = \{ \d \in D_n x^\alpha  \; | \;  \d \ast \Am \subseteq \Am \}.
\end{equation}

Now, using the fact that $ h_i \ast x^{\alpha} = (\alpha_i+1) x^{\alpha}$ and
for all $i=1, \ldots, n$ and  $\alpha \in \Z^n$,
we obtain the explicit expressions for the graded components,
$\CDAm_{[\alpha]}=\bigotimes_{i=1}^n \CD (A(m_i))_{[\alpha_i]}$, i.e.
$\CDAm =\bigotimes_{i=1}^n \CD (A(m_i))$.

2. Statement 2 follows from statement 1 and Theorem \ref{AB17Nov18}.(4).  $\Box$ \\

{\bf The algebra $\CDAm$ is a $\Z^n$-graded algebra.}  Recall that $\CDAm =\bigotimes_{i=1}^n\CD (A (m_i)) $ (Theorem \ref{GenRel30Jan20}.(1)).
If $m_i\geq 2$ then the algebra $\CD (A (m_i)) =\bigoplus_{j\in \Z}K[h_i]\d_j(i)$ is a $\Z$-graded algebra where the elements $\d_j(i)=\d_j$ are defined in Theorem \ref{AB18Nov18}.(1). If $m_i=1$ then the algebra $\CD (A(1))$ is the Weyl algebra $A_1=\bigoplus_{j\in \Z} K[h_i]\d_j(i)$ which is a $\Z$-graded algebra since it is a GWA where $\d_j(i)=\d_1(i)^j=x_i^j$ and $\d_{-j}(i)=\d_{-1}(i)^j=\der_i^j$ for $j\geq 0$. Since $\CDAm =\bigotimes_{i=1}^n\CD (A (m_i)) $ and every tensor component is a $\Z$-graded algebra the algebra $\CDAm$ is a $\Z^n$-graded algebra
%\marginpar{grCDAm}
\begin{equation}\label{grCDAm}
\CDAm =\bigotimes_{\alpha \in \Z^n} D_n\d_\alpha , \;\; D_n=K[h_1, \ldots , h_n], \;\; \d_\alpha =\prod_{i=1}^n\d_{\alpha_i}(i).
\end{equation}
Notice that $D_n\d_\alpha = \d_\alpha D_n$ since $\d_\alpha d= \s^\alpha (d)\d_\alpha$ where $\s^\alpha = \prod_{i=1}^n \s_i$, $\s_i(h_j) = h_j-\d_{ij}$. The $\Z^n$-grading on the algebra $\CDAm$ in (\ref{grCDAm}) coincides with the induced $\Z^n$-grading that is determined by the embedding $\CDAm\subseteq A_{n,x}$ and the $\Z^n$-grading of the algebra $A_{n,x}$ in (\ref{Anx}).  

{\bf The generalized Weyl algebras $\mA_n$ and $\mB_n$ such that $\mA_n \subset \CDAm \subset \mB_n \subset T_n^{-1}\mA_n =T_n^{-1} \CDAm = T_n^{-1} \mB_n $.} Recall that $\CDAm = \bigotimes_{i=1}^n \CD (A(m_i))$. For each number $i=1, \ldots , n$,   let $\mA (i)$ be the subalgebra of $\CD (A(m_i))$ which is generated by the elements $\d_{-1}(i)$, $h_i$ and $\d_1(i)$. By Theorem \ref{AB17Nov18}.(1),
 $\d_{-1}(i)= h_i(h_i-m)x_i^{-1}$ and $\d_1(i)= (h_i-2)x_i$, and so the algebra
%\marginpar{ADddn1}
\begin{equation}\label{ADddn1}
\mA (i) =D(i)[\d_1(i), \d_{-1}(i); \s_i , h_i(h_i-1)(h_i-m_i)], \;\; D(i)=K[h_i], \;\; \s_i (h_i) = h_i-1,
\end{equation}
is a GWA such that $\mA (i) \subset \CA_1(i) := \CD (A(m_i))\cap A_1(i)$ where $ A_1(i)=K\langle x_i , \der_i \, | \, \der_i x_i-x_i\der_i=1\rangle$ is the (first) Weyl algebra since $\d_{-1}(i), h_i , \d_1(i)\in  \CA_1(i)$ (Theorem \ref{BB18Nov18}.(2)). Let $$\mA_n:=\bigotimes_{i=1}^n \mA (i)\;\; {\rm and} \;\; \CA_n:=\bigotimes_{i=1}^n \CA_1 (i).$$
Then $\mA_n \subseteq \CA_n$.

The multiplicative submonoid $T(i)= \langle h_i-j\, | \, j\in \Z\rangle$ of $D(i)$ is a (left and right) denominator set of the algebras $\mA (i)$, $
\CA_1(i)$, $\CD (A(m_i))$ and $A_1(i)$ such that
%\marginpar{BBTn1}
\begin{equation}\label{BBTn1}
T(i)^{-1}\mA (i)\simeq T(i)^{-1}\CA_1 (i) \simeq T(i)^{-1} \CD (A(m_i)) \simeq  T(i)^{-1}A_1(i)=:  \mB (i)= T(i)^{-1}D(i)[x_i,x_i^{-1}; \s_i ]
\end{equation}
where $T(i)^{-1}D(i)=K[h_i, (h_i-j)^{-1}]_{j\in \Z}$ and $  \s_i (h_i) = h_i-1 $. Let
$$\mB_n := \bigotimes_{i=1}^n\mB (i)\;\; {\rm  and }\;\;\CA (m):=\bigotimes_{i=1}^n \CA (m_i)$$
 where $\CA (m_i)$ is a subalgebra of  $\CD (A (m_i))$ which is generated by the elements
$h_i$, $X_i:= x_i^{m_i}$ and $Y_i:= \d_{-m_i}(i)$. The algebra $\CA (m_i)$  is a  GWA of rank 1,
 $$\CA (m_i)= K[h_i][X_i,Y_i;\s_i^{m_i}, a_i=(h_i+m_i-1) \cdot (h_i-2) (h_i-3) \cdots (h_i-m_i)], $$ which  is a central simple  Noetherian domain
where $\s_i (h_i) = h_i-1$, see Theorem \ref{AB18Nov18}.(2).

We have the following diagram of algebras where the vertical lines denote containments of the algebras where $T_n:= T(1) \cdots T(n)$ is a denominator set of the corresponding algebras:
\begin{figure}[H]
\centering
\begin{tikzpicture}
\node (B)  at (0,2){$\mB_n= T^{-1}_n\mA_n = T^{-1}_n\CA_n  = T^{-1}_n \CD (A(m)) =  T^{-1}_nA_n = T^{-1}_nD_n[x_1^{\pm 1}, \ldots ,x_n^{\pm 1}; \s_1, \ldots, \s_n]$};
\node (C) at (-1,1) {$\CD(A(m))$};
\node (D)  at (1,1){$A_n$};
\node (E)   at (0, 0){ \qquad \qquad \qquad \qquad  $\CA_n=\CD (A(m) \cap A_n)$};
\node (F)   at (-1, 0){$\CA(m)$};
\node (G)   at (0,-1){$\mA_n$};
\draw [semithick, -] (B)--(C);
\draw [semithick, -] (B)--(D);
\draw [semithick, -] (C)--(F);
\draw [semithick, -] (E)--(G);
\draw [semithick, -] (C)--(E);
\draw [semithick, -] (D)--(E);
\end{tikzpicture}
\caption{AADBB1} \label{AADBB1}
\end{figure}


\begin{proposition}\label{A10Jan20}
%\marginpar{A10Jan20}
Let $m=(m_1, \ldots ,m_n)\in \N^n$. Then
\begin{enumerate}
%\item The algebra $\CD (A)$ is a central simple  Noetherian domain.
\item The subalgebra $\CA (m)$ of $\CDAm$ is a  GWA of rank $n$ which  is a central simple  Noetherian domain of Gelfand-Kirillov dimension $2n$.
\item The algebra $\CDAm$ is a finitely generated  left and right $\CA (m)$-module,
$$ \CDAm  = \sum_{ \{ \alpha \in \Z^n : |\alpha_1| < 2m_1, \ldots ,  |\alpha_n| < 2m_n \} } \CA (m) \d_\alpha  = \sum_{ \{ \alpha \in \Z^n : |\alpha_1| < 2m_1, \ldots ,  |\alpha_n| < 2m_n \} } \d_\alpha  \CA (m). $$
\end{enumerate}
\end{proposition}

{\it Proof}. 2. Statement 2 follows from the fact that $\CDAm =\otimes_{i=1}^n\CD (A(m_i))$ and Theorem \ref{AB18Nov18}.(3).

1. By \cite[Proposition 1.3]{Bav-GWArep}, the GWA $\CA (m)$ a   Noetherian domain. By \cite[Theorem 4.5]{Bav-FilDimSimCrit-1996}, the GWA $\CA (m)$ a  simple algebra. The algebra $\CA (m)$ is central since the algebra $A_{n,x}$ is so and
$$\CA (m) \subset S_{n,x^m}^{-1}\CA (m) \simeq D_n[x_1^{\pm m_i}, \ldots , x_n^{\pm m_n}; \s_1^{m_1}, \ldots , \s_n^{m_n}]\simeq A_{n,x}\;\; (x_i^{m_i}\mapsto x_i, \;\; h_i\mapsto m_i h_i).$$
 The GWA $\CA (m)=\otimes_{i=1}^n\CA (m_i)$ is a tensor product of simple GWAs (see Theorem \ref{AB18Nov18}.(2)). By \cite[Corollary 4.8.(2)]{Bav-FilDimSimCrit-1996}, the Gelfand-Kirillov dimension of the algebra $\CA (m)$ is $2n$.  Now, by statement 2 and \cite[Propositiion 8.2.9.(ii)]{MR}, $\GK (\CDAm ) = \GK (\CA (m))=2n$. $\Box$ \\


{\bf Proof of Theorem \ref{GenProp30Jan20}.}

 (i) {\em The algebra $\CDAm$ is central}: $K\subseteq Z(\CDAm )\stackrel{ (\ref{ASAD1n1})}{\subseteq} Z(A_{n,x})=K$, and so the  algebra $\CDAm$ is central.

(ii) {\em The algebra $\CDAm$ is Noetherian with Gelfand-Kirillov dimension $2n$}: By Proposition \ref{A10Jan20}, the subalgebra $\CA (m)$ of $\CDAm$  is a Noetherian algebra of  Gelfand-Kirillov dimension $2n$ such that the algebra $\CDAm$ is a finitely generated left and right $\CA (m)$-module. Hence, the algebra $\CDAm$ is also Noetherian and by \cite[Proposition 8.2.9.(ii)]{MR},  $\GK (\CDAm ) = \GK (\CA (m))=2n$.

(iii) {\em The algebra $\CDAm$ is simple and $\Z^n$-graded}: By (\ref{grCDAm}), the algebra $\CDAm$ is a $\Z^n$-graded algebra and the $\Z^n$-graded components $D_n\d_\alpha $ ($\alpha \in \Z^n$) of the algebra $\CDAm$ are the common eigenspaces of the commuting inner derivations $\ad_{h_1}, \ldots , \ad_{h_n}$ of the algebra $\CDAm$. Therefore every nonzero ideal of the algebra $\CDAm$ is a homogeneous ideal and as a result  has nontrivial intersection with the subalgebra $D_n$ of $\CDAm$. Since $D_n\subseteq \CA (m)$ and the algebra $\CA (m)$ is simple (Proposition \ref{A10Jan20}.(1)), all nonzero ideals of the algebra $\CDAm$ are equal to $\CDAm$, and so the algebra $\CDAm$ is a simple algebra.  $\Box$ 

{\bf The Krull dimension of the algebras $\CD (\Am )$.} {\bf Proof of Theorem \ref{Kdim30JAn20}.}

 By \cite[Corollary 4.8.(5)]{Bav-FilDimSimCrit-1996}, the Krull dimension of the GWA $\CA (m)$ is $n$.  By Proposition \ref{A10Jan20}, the algebra $\CDAm$ is  a finitely generated left and right $\CA (m)$-module. Hence, the Krull dimension of the algebra $\CDAm$ is smaller or  equal to the Krull dimension of the algebra $\CA (m)$ which is $n$. The polynomial algebra $D_n$ is the zero graded component of the $\Z^n$-graded algebra $\CDAm$. Hence, them map $I\mapsto \CDAm\t_{D_n}I$  (resp., $I\mapsto I\t_{D_n} \CDAm$) from the set of ideals of the algebra $D_n$ to the set of left (resp., right) ideals of the algebra $\CDAm$ is an injection. Hence,  the Krull dimension of $D_n$, which is $n$,  is smaller or equal to the Krull dimension of $\CDAm$. Therefore, the Krull dimension of the algebra $\CDAm$ is $n$. $\Box$

{\bf An analogue of the Inequality of Bernstein for the algebras  $\CD (\Am )$.}  By \cite[Corollary 4.8.(4)]{Bav-FilDimSimCrit-1996}, an analogue of the  the Inequality of Bernstein holds for the algebra $\CA (m)$: {\em For  all nonzero finitely generated $\CA (m)$-modules} $M$, $\GK (M)\geq n$.

{\bf Proof of Theorem \ref{BerIn30JAn20}.}  By Proposition \ref{A10Jan20}, the algebra $\CDAm$ is  a finitely generated left and right $\CA (m)$-module.  Hence, each finitely generated $\CDAm$-module $M$ is a also a nonzero finitely generated $\CA (m)$-module. Now,
 $$ \GK_{\CDAm}(M)\geq  \GK_{\CA (m)}(M)\geq n,$$
and Theorem \ref{BerIn30JAn20} follows.  $\Box$

{\bf The global dimension of the algebras $\CD (\Am )$.} Recall that Morita equivalent algebras have the same global dimension and  the global  dimension of the Weyl algebra $A_n$ is $n$ (in characteristic zero). 

{\bf Proof of Theorem \ref{gldim30JAn20}.}  By \cite[Theorem, p.29]{Musson-1986}, in the case $n=1$, the algebra $\CD (A(m_1))$ is Morita equivalent to the Weyl algebra $A_1$. Hence, for an arbitrary $n\geq 1$, the algebra $$\CD (A(m))=\CD (\otimes_{i=1}^nA(m_i))\simeq \otimes_{i=1}^n\CD (A(m_i))\;\; {\rm (Theorem \ref{GenProp30Jan20}.(1))},$$  where $m\in \N^n$, is Morita equivalent to the Weyl algebra $A_n=A_1^{\otimes n}$.
 Therefore, the global dimension of the algebra $\CD (A(m))$ is equal to the global dimension of the Weyl algebra $A_n$ which is $n$. $\Box$


\end{document}
